The Experiences of Teaching Portfolio Preparation and Maintenance by Faculty

by

Byron Matthews

A thesis submitted in partial fulfillment of the requirements for the degree of

Master of Education

in

Educational Administration and Leadership

Department of Educational Policy Studies
University of Alberta

© Byron Matthews, 2018

Abstract

Teaching portfolios have been used by faculty in postsecondary institutions for summative and formative evaluations since the early 1980s (Knapper, 1995). Despite their proliferation, there remains much to be learned about the use of teaching portfolios and whether they are perceived as being effective for formative and/or summative evaluations. To expand on our understanding of teaching portfolio use by faculty, the researcher used a generic qualitative study (Caelli, Ray, & Mill, 2003) to explore the process in which faculty prepare and maintain a teaching portfolio and the perceived effect that teaching portfolios have on summative and formative evaluations. A theoretical framework of self-directed learning (SDL) (Brookfield, 1985; Garrison, 1997; Hiemstra, 1994; Knowles, 1975) was used as a guide to explore teaching portfolios. The results of the study reveal that teaching portfolios are a SDL activity that are prepared in many ways, used for a variety of purposes, and are perceived by faculty as having varying degrees of effectiveness for summative and formative evaluations. Drawing from the results of this study, five main conclusions can be made: 1) faculty teaching portfolios are a SDL tool; 2) a wide range of support and resources are found to be valued by faculty when preparing a teaching portfolio; 3) teaching portfolios vary in content; 4) teaching portfolios are initially prepared for numerous reasons and are used for a variety of purposes; 5) the effectiveness of teaching portfolios as a summative and/or formative evaluation activity vary among faculty.

There are a number of implications that can be drawn from these main conclusions.

Firstly, as the process of preparing a teaching portfolio is a SDL activity, resources should be available for faculty who are preparing a teaching portfolio and clearly communicated to faculty. Secondly, study participants indicated a wide range of items they have used in their teaching portfolios. Faculty can use these items as a resource from which to choose the content of their

teaching portfolio from. Likewise, this study has concluded that teaching portfolios are used for a wide range of purposes – information that faculty may find useful when deciding if preparing a teaching portfolio is a useful endeavour. Lastly, the effectiveness of a teaching portfolio as a summative and formative evaluation tool is mixed, indicating that caution must be used when using them for various forms of evaluation.

Preface

This thesis is an original work by Byron Matthews. The research project related to this thesis received research ethics approval from the University of Alberta Research Ethics Board, Project Name "A Study of the Experiences of Teaching Portfolio Preparation and Maintenance by University of Alberta Faculty," No. Pro00053249, June 9, 2015.

Acknowledgements

Thank you to my supervisor, Dr. Heather Kanuka, for her continued support throughout the process of writing this thesis. This thesis would not have been possible without her unfailing guidance and positivity.

A very special thank you to my family, especially my wife, who encouraged me throughout the process of writing this thesis. Their love and patience was invaluable, and I cannot express how much I appreciate their endless support.

Table of Contents

Abstract	ii
Preface	iv
Acknowledgements	v
Chapter One: Introduction	1
Overview	1
Context of the Study	2
Background and Existing Research	4
Need for the Study	5
Statement of the Problem	5
Research Questions	5
Theoretical Framework Guiding the Study	6
Limitations	7
Delimitations	8
Definition of Terms	9
Teaching portfolio	9
E-Portfolio	9
Summative evaluation	9
Formative evaluation	9
Postsecondary education	9
Faculty member	
Tenure	
Self-directed learning	
Organization of the Thesis	11
Chapter Two: Related Literature	12
Overview	
Self-Directed Learning Literature	
History of Self-Directed Learning	
Description of Self-Directed Learning	
The Inquiring Mind	
Results	21

Learner Subgroups	22
Summary	24
Teaching Portfolio Literature	24
Introduction	24
Teaching Portfolio Description	25
Teaching Portfolio Content	26
Summative Use of Teaching Portfolios	29
Formative Use of Teaching Portfolios	31
Benefits of Teaching Portfolio Use	32
Concerns of Teaching Portfolio Use	34
Chapter Summary	37
Chapter Three: Methods	38
Overview	38
Research Paradigm	38
Methodology	39
Theoretical Positioning	40
Methods	41
Rigor	41
The Analytic Lens	42
Study Site	42
Participant Recruitment	42
Participants	43
Data Collection	43
Data Analysis	44
Trustworthiness	46
Generalizability	46
Chapter Summary	47
Chapter Four: Results	48
Overview	48
Participant Demographics	48
Teaching Portfolio Preparation and Maintenance	50
Teaching Portfolio Preparation Guidance and Resources	51

Teaching Portfolio Content	56
Teaching Portfolio Components Perceived as Beneficial	60
Teaching Portfolio Components Perceived as Being Least Helpful	62
Teaching Portfolio Uses	63
Teaching Portfolio Format	65
Teaching Portfolio Maintenance	69
Teaching Portfolio Encouragement	71
Teaching Portfolio Effectiveness	72
Do Teaching Portfolios Improve Teaching?	74
Teaching Portfolios for Formative Evaluation	78
Teaching Portfolios for Summative Evaluation	80
Are Teaching Portfolios an Effective Use of Time?	83
Would You Recommend a Teaching Portfolio to a Colleague?	85
Chapter Summary	88
Chapter Five: Discussion and Implications	89
Overview	89
Teaching Portfolios as a Self-Directed Learning Activity	89
Teaching Portfolios and Self-Directed Learners	91
Teaching Portfolio Preparation and Maintenance	93
Teaching Portfolio Content	94
Teaching Portfolio Uses	95
The Perceived Effectiveness of Teaching Portfolio Use	96
Teaching Portfolios for Formative Evaluation	98
Chapter Summary	101
Chapter Six: Reflections and Recommendations	102
Overview	102
Recommendations for Future Study	103
References.	105
Appendix A: Department Chair Email	112
Appendix B: Letter of Invitation	113
Appendix C: Participant Consent	115
Appendix D: Interview Questions	117

Chapter One: Introduction

Overview

This thesis originated out of a desire to learn more about methods that are currently used by faculty to improve their teaching. As an instructor in a postsecondary institution, I am always keen to discover new ways to improve the learning experience that I provide to my students.

Early on in my studies as a graduate, student I discovered teaching portfolios and since then, I have been motivated to learn more about how they are used by faculty and postsecondary institutions as a summative and formative evaluation tool. I believe that by exploring how others have used a teaching portfolio, it is possible to provide additional insights on how to use them effectively. Similar to most teachers, I endeavor to continuously improve as a teacher. Gaining insights on how to effectively use teaching portfolios is part of that process.

An assumption I hold, and one that underpins this study, is that teaching portfolios can have a positive effect on teaching and learning. Faculty in postsecondary institutions play a critical role in helping students form a world view. From leading a large lecture class, to facilitating labs and seminars, to supervising graduate students, to teaching online courses – the interactions between faculty and students are varied and immense. It is through these interactions that students learn to be engineers, scientists, doctors, artists, teachers – the list goes on. However, it is currently impossible (and may always be) to fully appreciate the effects of these interactions. We do know that some interactions may be beneficial while others are not. Regardless, it is of utmost importance that we strive to understand the complex relationships between students and faculty so that they are positive and beneficial. One way we do this as teachers is by reflecting on our teaching, and in doing so, evaluate our teaching. The use of

teaching portfolios has been viewed as an effective way to evaluate our teaching through formative and summative evaluation (Seldin, Miller & Seldin, 2010).

Context of the Study

The use of teaching portfolios is an established activity by teachers for summative and formative evaluation in colleges and universities both nationally and internationally (Seldin, Miller & Seldin, 2010). While there is a good body of research on how to use teaching portfolios, there is rather scant research on their perceived effectiveness by those who use them. Gaining further insights on how teachers perceive the effectiveness of teaching portfolios for formative and summative evaluation can make it possible to enhance teaching practices and, *de facto*, the learning process.

Most university faculty members develop their own teaching portfolios using some form of SDL. SDL is a form of learning that provides the learner with control of what is learned, how it is learned, and for what purposes. Supporting this notion, Heimstra (1994) indicated that SDL is a study form whereby individuals have primary responsibility for planning, implementing, and evaluating effort. Similarly, Garrison (1997) describes SDL "as an approach where learners are motivated to assume personal responsibility and collaborative control of the cognitive (self-monitoring) and contextual (self-management) processes in constructing and confirming meaningful and worthwhile learning outcomes" (p. 18). Although SDL may be expressed in a variety of ways and may have different outcomes, what is clear is that it is a form of learning whereby the learner has some degree of control in the learning process.

The use of self-directed teaching portfolios in postsecondary institutions across North America has been used since the 1970s (Knapper, 1995). Also known as teaching dossiers, the

Canadian Association of University Teachers (2007) provides the following description of teaching dossiers:

A teaching dossier is a summary of an academic's major teaching accomplishments and strengths. It is to an academic's teaching what lists of publications, grants, and academic honours are to research. The teaching dossier is intended to provide short statements which describe the scope and quality of the academic's teaching. (p. 2)

Seldin, Miller and Seldin (2010) provide a similar description of teaching portfolios:

[A] teaching portfolio is a factual description of a professor's teaching strengths and accomplishments. It includes documents and materials that collectively suggest the scope and quality of a professor's teaching performance. The portfolio is to teaching what lists of publications, grants, and honors are to research and scholarships. (p. 4)

While teaching portfolios have been used for many purposes, Urbach (1992) contends "the goal of a teaching portfolio is to describe, through documentation over an extended period of time, the full range of your abilities as a college teacher" (p. 71). Additionally, scholars have frequently noted that teaching portfolios are used for summative (Berk, 2005; Buckridge, 2008; Knapper, 1995; Redmon, 1999; Seldin, Miller and Seldin, 2010; Weeks, 1996) and formative (Kaplan, 1998; Reece, Pearce, Melillo, & Beaudry 2001; Retallick, 2000; Seldin, Miller and Seldin, 2010) evaluation of faculty. Based on these definitions, teaching portfolios can be understood as a means to document teaching activities and are used for numerous purposes.

Since their beginnings, the use of teaching portfolios at postsecondary institutions has expanded. Almost two decades ago, Seldin (2000) estimated that more than 1500 colleges and universities in the United States and Canada used teaching portfolios for various purposes.

Around the same point in time, Knapper (1995) also indicated that teaching portfolios were

extensively in use for some time, stating that "today teaching portfolios are known on nearly every college and university campus in the United States" (p. 45). More recently, in Canada, Gravestock (2011) identified 18 Canadian postsecondary institutions that require teaching portfolios for tenure and another 11 that recommend them or accept them as optional sources of data provided by the instructor for tenure applications. With the current use and likely continued growth of teaching portfolios, it is important that we have a good understanding of how they are used and whether faculty believe they are effective for summative and/or formative evaluation.

Background and Existing Research

While teaching portfolios may be developed in many ways, Trevitt and Stocks (2012) indicate that they tend to consist of two main parts — documentation provided by the candidate as 'evidence' of their practice and a narrative commentary which offers a rationale for the approach evidenced. In other words, teaching portfolios contain *what* material is used by the teacher, *why* that material was chosen, and *how* the material was used. However, specific components of teaching portfolios can vary considerably (Corry & Timmins, 2009; Simpson, Morzinski, & Lindemann, 1994; Urbach, 1992), and are usually unique to the individual who creates the portfolio (Reece et al., 2001; Seldin, 2000; Seldin, Miller & Seldin, 2010), which can result in difficulties when they are used for various forms of evaluation. Additionally, Oermann (1999) suggested that the content depends on whether a teaching portfolio is used for formative or summative evaluation. These differences may influence the way teaching portfolios are used as a form of SDL, and their effectiveness for formative or summative evaluation may be questioned.

While many researchers have indicated that teaching portfolios can be used for summative evaluation (Berk, 2005; Buckridge, 2008; Knapper, 1995; Redmon, 1999; Seldin,

Miller & Seldin, 2010; Weeks, 1996) and/or formative evaluation (Kaplan, 1998; Reece et al., 2001; Retallick, 2000; Seldin, Miller & Seldin, 2010), there are some who question their effectiveness for these purposes (Dinham & Scott, 2003; Wolf, 1996; Wright, Knight, & Pomerleau, 1999). As such, postsecondary institutions may question the credibility of teaching portfolios for summative and/or formative evaluations. Additionally, faculty need to be provided with evidence that the time and effort they dedicate to preparing and maintaining a teaching portfolio is time well spent. Most assuredly, care must be taken when determining the content, format, and purpose of the portfolio. There is general agreement among scholars that teaching portfolios can be an excellent form of summative and/or formative evaluation; however, this view is not universal.

Need for the Study

With possible problems inherent in teaching portfolio use, it would be beneficial for those using teaching portfolios to have an evidence-based understanding of how they are used and whether they are perceived by faculty as an effective form of summative and/or formative evaluation. Currently there is limited research exploring the experiences and perceptions of faculty when preparing and maintaining a teaching portfolio.

Statement of the Problem

The purpose of this qualitative research project was to gain greater insights into the experiences that faculty members in postsecondary education institutions have had preparing and maintaining their self-directed teaching portfolio for the purpose of teaching improvement.

Research Questions

This study was guided by the following primary research question:

In what ways do faculty in postsecondary institutions prepare and perceive the use of teaching portfolios as an effective form of self-directed learning to improve their teaching?

Guiding the research question are the following sub-questions:

- a. In what ways do faculty prepare and maintain their teaching portfolio?
- b. How do faculty use support and resources to guide their teaching portfolio preparation?
- c. What are the key components that comprise their teaching portfolios?
- d. In what ways (if any) do faculty use teaching portfolios for summative and/or formative purposes?
- e. In what ways (if any) do faculty perceive teaching portfolios as improving their teaching?

Theoretical Framework Guiding the Study

Houle's (1961) seminal work on SDL suggests that adult learners generally fit within the following three categories when describing their reasons for participating in learning activities:

(a) goal-oriented, who participate mainly to achieve some end goal; (b) activity-oriented, who participate for social or fellowship reasons; (c) learning-oriented, who perceive of learning as an end in itself. The findings of this study indicate the participants exhibited characteristics of one or more of these categories when preparing and/or maintaining a teaching portfolio. While there is ample research exploring other methods of SDL, there is none that specifically describes teaching portfolios as a form of SDL. Using the research of Houle (1961) as a framework, this research will help to fill this gap.

Limitations

This study has limitations which generally fall into the following categories: (a) the generalizability of the findings; (b) the number of participants and participant demographics; (c) the accuracy of participant responses; (d) researcher bias related to the analysis of the data and subsequent conclusions.

Teaching portfolios are not used universally across or within postsecondary institutions. As such, research exploring teaching portfolio use is only applicable to faculty and institutions that currently use teaching portfolios or those who intend to use teaching portfolios in the future. The results of this study help to inform their use but are not meant to be used as prescriptive guidelines. While the findings of this study can assist postsecondary institutions and faculty alike, they cannot be generalized to all situations.

The number of faculty members who participated is another limitation of the study. Generalizations based on small sample sizes are potentially inaccurate and need to be interpreted as such. Likewise, the demographics of the study participants limits the transferability of the results of the study. Participant recruitment was intended to attract participants from all departments at the University of Alberta. However, it was likely that many faculty members were unaware of the study, and therefore did not have the opportunity to participate. Participant recruitment was facilitated through department chairs, so if a chair elected not to distribute the details of the study to faculty in their department, it is unlikely that those faculty would have been aware of the opportunity to participate. As such, the participants were not a representative sample of the University of Alberta faculty.

Another potential limitation of this study was the accuracy of the participants' responses. Specifically, some participants may have had biases – whether positive or negative – about

teaching portfolio use that is outside the normal populations' experiences. Therefore, the participants of the study, arising from those who volunteered to participate, may have consisted of those with strong opinions rather than a random sample which could have resulted in a different range of responses. Also, the participants of the study were asked questions that required them to recall experiences that were months or years in the past. As such, there is likely to be inherent inaccuracy of the responses.

Lastly, researcher bias was a limitation in the study as a result of the selection and wording of the interview questions, the analysis of the results, and the conclusions made in the findings. While there was a strong desire to limit biases through the use of traditional trustworthiness practices in qualitative research (see Chapter 4, Data Analysis), it was impossible to fully bracket researcher biases.

Delimitations

Delimitations of the study included the method of the data collection chosen and the participant selection process.

While the participant interviews elicited detailed and descriptive responses, the interviews were limited to 30 minutes to an hour. This decision was reached in the hopes that recruitment of participants would be more effective if the interviews were limited to an hour. It is possible, however, that longer interviews would have provided deeper insights into the process of teaching portfolio preparation and maintenance.

Another delimitation of this study was the participant selection process. Participant recruitment was facilitated by using a participant 'Letter of Invitation' (see Appendix A) distributed to University of Alberta department Chairs and Associate Chairs. This method of recruitment was chosen because it was an efficient way to connect with faculty members across

the University of Alberta in a short period of time. However, it is possible that a more extensive recruitment process (i.e. at other institutions) would have resulted in more, and diverse, participants.

Definition of Terms

This section provides descriptions and definitions of important terms that were used in this study.

Teaching portfolio

The Canadian Association of University Teachers (2007) describes teaching dossiers as "a summary of an academic's major teaching accomplishments and strengths. It is to an academic's teaching what lists of publications, grants, and academic honours are to research. The teaching dossier is intended to provide short statements which describe the scope and quality of the academic's teaching" (p. 2).

E-Portfolio

Barrett (2007) states that "an electronic portfolio uses technologies as the container, allowing students or teachers to collect and organize portfolio artifacts in many media types (audio, video, graphics, text)" (p. 438).

Summative evaluation

Summative evaluations are evaluations that are done at the end of an activity, process, or period of time in order to understand if goals or objectives were met (Mertens, 2010).

Formative evaluation

Formative evaluations are defined in this study as evaluations conducted "during the operation of a program to provide information useful in improving the program" (Mertens, 2010, p. 71).

Postsecondary education

Statistics Canada (2010) indicates that

Postsecondary education refers to those whose highest level of educational attainment is an apprenticeship or trades certificate or diploma (including 'centres de formation professionnelle'); college, CEGEP (Collège d'enseignement général et professionnel) or other non-university certificate or diploma; university certificate or diploma below bachelor level; or a university degree (bachelor's degree; university certificate or diploma above bachelor level; decree in medicine, dentistry, veterinary medicine or optometry; master's degree; earned doctorate).

Faculty member

Manning (2013) describes faculty members as individuals working in postsecondary education who have various titles (e.g. adjunct, clinical, research), are ranked as either assistant, associate, and full professor, and have teaching, research, and service responsibilities.

Tenure

The University of Alberta Faculty Agreement defines tenure as "an appointment to the academic staff without term, which may be terminated only by resignation, retirement, death, or in accordance with the terms of this Agreement" (University of Alberta, 2006, p. 2)

Self-directed learning

Knowles (1975) defines self-directed learning: "In its broadest meaning, "self-directed learning" describes a process in which individuals take the initiative, with or without the help of others, in diagnosing their learning needs, formulating learning goals, identifying human and material resources for learning, choosing and implementing appropriate learning strategies, and evaluating learning outcomes" (p. 18).

Organization of the Thesis

This chapter provided an introduction to the study, discussed the background and existing research related to the study, explained the need for the study, identified the study problem and related research questions, identified the theoretical framework guiding the study, stated the significance of the study, presented limitations and delimitations, and defined important terms used in this study. Chapter 2 provides a comprehensive review of the literature of SDL and teaching portfolios as they relate to this study. Chapter 3 presents the research paradigm of this thesis, explains the research framework guiding the study, discusses the study research design and methodology, describes the participants and participant recruitment process, and indicates the data collection and analysis processes. Chapter 4 presents the results of the study. Chapter 5 provides a thorough discussion of the study findings and possible implications. Chapter 6 concludes the thesis by presenting researcher reflections and discussing recommendations for future research.

Chapter Two: Related Literature

Overview

The previous chapter provided a brief description of why I decided to do this research, introduced key concepts of the study, presented background and existing research related to the study, explained the need for the study, discussed the study's problem statement, and identified the research questions guiding this study. It also described the theoretical framework guiding the study and the need for the study. Chapter 1 also presented the limitations and delimitations of this study as well as definitions of terms. Lastly, the previous chapter outlined the organization of the thesis.

This chapter includes a review of the literature related to this study. SDL literature is explored, as is the literature of teaching portfolios. This chapter provides a discussion of the context in which this study is founded and provides the concepts guiding this study.

Self-Directed Learning Literature

Introduction

SDL has been extensively studied by numerous academics for decades. The complexities of SDL have provided a rich source of research questions that have been examined from numerous perspectives. However, there remains much to be explored in this area. In particular, the effectiveness of SDL has been underexplored with respect to various contexts. While it is difficult to understand how SDL is used in all contexts, it is useful to study in the specific contexts in which it is frequently used. The extensive use of teaching portfolios among postsecondary education faculty members as a form of SDL for summative and formative evaluation necessitates this as an area warranting further exploration.

The following review of the literature on SDL provides a summary of what SDL is and how it relates to the use of teaching portfolios by faculty. Lastly, this section of the literature review discusses why and how the work of Houle (1961) was used as the theoretical framework for this study.

History of Self-Directed Learning

Throughout history people have sought out ways to increase their ability to learn from the world around them. Learning occurs in numerous formats that involve other people including mentorships, discussions groups, and group classes. In many cases, individuals prefer another type of learning environment – one in which much of the control of learning lies in the hands of the individual. SDL is an example of this type of learning.

While most of us have at some time used SDL, it has only really been an area of research for less than a century. When referring to SDL, for example, Caffarella (1993) indicated that "the first contribution has its roots in the work of Houle (1961)" (p. 27). Numerous other scholars have since noted the significance of Houle's influence on the study of SDL. Hiemstra (1994) also notes that much of the groundwork on SDL was provided by Houles' research in 1961. Likewise, Brockett and Donaghy (2011) note that "Cyril Houle is frequently credited as having played a pivotal role in bringing self-directed learning (or self-education as he frequently called it) to the forefront of scholarship in adult education and learning" (p. 1). Research by Confessore and Confessore (1992) who surveyed experts on SDL indicated that *The Inquiring Mind* (1961) is one of "the most important works that, in the panel's judgement, should be read at the outset of one's introduction to the field of adult self-directed learning" (p. 17). When referring to Houle's book *The Inquiring Mind*, Guglielmino, Long, and Hiemstra (2004) stated that "one of the most influential research studies identified with self-direction in learning in the

United States was reported in 1961 by Cyril Houle" (p. 4). It is clear that Houle's research has had a significant effect on the study of SDL. However, research into this area of study did not end there; in fact, it has been ongoing for more than half a century.

The work done by Houle in 1961 sparked a line of research into SDL that has endured. In the decades that followed, Tough (1978) and Knowles (1975) explored the area of SDL and have also been considered leading scholars in the field ever since. Merriam (2001) highlights the influence of these pioneering SDL scholars and stated that "based on the pioneering work of Houle, Tough, and Knowles, early research in self-directed learning was descriptive, verifying the widespread presence of self-directed learning among adults and documenting the process by which it occurred" (p. 8). Over the last few decades, research into SDL became a significant area of study. For example, Brockett et al. (2000) note that "few topics have commanded as much attention in the literature of adult education, over the past several decades, as self-directed learning" (p. 3). Likewise, Brockett and Donaghy (2011) indicated that "since the early 1970s, self-directed learning (SDL) has emerged as one of the most influential areas of study and practice in adult education" (p. 1). Hiemstra (1994) also noted that SDL research and scholarship is significant and that it may be one of the most explored topics in adult education. Guglielmino, Long, and Hiemstra (2004) have noted that

[S]elf-direction in learning has been one of the most active streams of inquiry in adult education research in the US in the past 40 years and the attention to self-direction in learning, both in the United States and internationally, is unlikely to diminish. (p. 11) Mezirow (1985) highlighted the role of SDL research in our understanding of adult learning by asserting that "no concept is more central to what adult education is all about than self-directed

learning" (p. 17). Likewise, Garrison (1997) described the significance of SDL as a topic in the field of adult education, stating

One important reason has to be the intuitively appealing desire to be in control of deciding what to learn and how to learn it. It also fits with the desire and need felt by most adults to continue to learn. These innately human characteristics are inherent in the concept of self-directed learning. (p. 19)

Caffarella (1993) reinforces the importance of research in SDL by suggesting that "the concept has an almost cult like quality to the extent that self-directedness is viewed as the essence of what adult learning is all about" (p. 25).

Although there is a clear consensus among scholars that SDL is an important area of study, there is no on one definition or description of SDL. Additionally, there are many contexts in which SDL is used that are not fully understood. One purpose of this study is to explore one of these contexts to expand our understanding of SDL.

Description of Self-Directed Learning

Research into the area of SDL has occurred for many decades and has explored the topic from a multitude of vantage points. As touched on above, there is no one description or definition of SDL that has been agreed upon by scholars. However, there are recurrent themes. To better understand the impact of SDL, it is important to review some concepts relating to this area of study.

There is agreement among scholars that one key component of SDL is the amount of control the learner has over the learning environment. Hiemstra (1994), for example, noted that SDL includes any study form whereby the individual has the primary responsibility for planning, implementing, and evaluating the effort. Likewise, Brookfield (1985) highlights the importance

of individual control by indicating that "individual control over learning is often claimed to be the distinctive characteristic of self-directed learning" (p. 7). Garrison (1997) reinforces this position by stating that "the apparent need to "learn on one's own" has been a persistent theme in self-directed learning" (p. 19). Lastly, Guglielmino, Long, and Hiemstra (2004, p. 5) indicated that the following definition of SDL provided by Knowles (1975) is the best-known and most-cited:

In its broadest meaning, "self-directed learning" describes a process in which individuals take the initiative, with or without the help of others, in diagnosing their learning needs, formulating learning goals, identifying human and material resources for learning, choosing and implementing appropriate learning strategies, and evaluating learning outcomes. (p. 18)

This definition indicates that learners are responsible for much of their learning environment. It is reasonable to conclude that a significant degree of control of ones' learning is a necessary aspect of SDL.

Although research into SDL has provided us with a rich source of information to draw from, there still is uncertainty as to when this type of learning is best used. It has been well documented in the literature that SDL is a desired form of learning for adults. Guglielmino, Long, and Hiemstra (2004) stated, for example, that "it is the contention of the authors that learner self-direction is a universal human attribute; it is present in each person to some degree" (p. 1). However, it has also been pointed out that while SDL has its strengths, it isn't appropriate for all individuals or all learning situations. Supporting this notion, Guglielmino, Long, and Hiemstra (2004) suggested:

... although certain learning situations are more conducive to self-direction in learning than are others, it is the personal characteristics of the learner – including his or her qualities of mind and behavior (personality) as well as acquired skills and abilities – which ultimately determine whether self-directed learning will take place in a given learning situation. (p. 1)

Brookfield (2009) also identified the importance of understanding each learner as they relate to SDL. Additionally, Brookfield (2009) noted the importance of collaborating with others while using SDL:

[I]ndeed, a recurring theme of research in this area is the way learners move in and out of learning networks and consult a range of peers. The key point is that whether or not learners choose to be temporarily isolated from, or immersed within, peer networks is the learner's decision. (p. 2615)

Arising from the prior literature, SDL appears to be effective when the learners' characteristics and environmental factors are taken into account. In some instances learners may choose to learn independently, while in other situations collaboration and guidance with others may be preferred.

Although differences in learner characteristics and environmental factors need to be considered when using SDL, it has been suggested that SDL can be effective in a variety of situations. Guiglielmino, Long, and Hiemstra (2004) supported this assertion: "self-direction in learning can occur in a wide variety of situations, ranging from a teacher-directed classroom to self-planned and self-conducted learning projects" (p. 1). Brookfield (2009) also indicated that SDL does not need to be done in isolation and that, in fact, "learners can work in self-directed ways while engaged in group-learning settings, provided that this is a choice they have made

believing it to be conducive to their learning efforts" (p. 2615). There are clearly two key points to reiterate about Brookfield's position: 1) that SDL occurs in numerous settings and 2) the learner chooses what setting to learn in. Hiemstra (1994) also supported a similar position suggesting that the conditions for optimal learning occur when a learner's level of self-direction is supplemented with possible SDL opportunities. In other words, SDL is effective when the learner is aware of potential sources of support and has the power to choose from them.

In addition to being a preferential form of learning for many, SDL has numerous other benefits which have been identified in the literature. Firstly, it has been suggested that SDL can lead to a greater awareness of one's learning characteristics and preferences. Reinforcing this point, Garrison (1997) noted that "opportunities for self-directed learning, in turn, enhance metacognitive awareness and create the conditions where students learn how to learn" (p. 31). This increased awareness has the potential to improve other learning situations outside of SDL. Related to this aspect of awareness, Confessore and Kops (1998) suggested that "self-directed individuals reflect, assess, and evaluate rather than uncritically accept and internalize information" (p. 365). This process of learning, reflecting, and evaluating enhances self-awareness and can also contribute to improved learning. When looking at the broader effects of SDL, Caffarella (1993) summarized its benefits:

[S]elf-directed learning has contributed to our understanding of learning by (1) identifying an important form of adult learning and providing us with insights into the process of learning, (2) challenging us to define and debate the salient characteristics of adult learners, and (3) expanding our thinking about learning in formal settings. (p. 27) Essentially, SDL can enhance the learning opportunities of individuals, and research into SDL has expanded our overall understanding of adult learning.

Although there is a rich history of research related to SDL, there continues to be much we can learn. For example, previous research on SDL has tended to be qualitative in nature, limiting the generalizability of the findings. It is also important to explore SDL in various settings so we can improve our understanding of its effectiveness in various contexts. There are a multitude of frameworks from which to view SDL. One way has been to examine this process through the characteristics of the learners. For example, Houle's (1961) work provided one way to categorize learners based on specific learner characteristics. As this study builds on the work of Houle (1961), it is important to describe some of his research on adult learners as it relates to this study.

The Inquiring Mind

Although it is understood that SDL was occurring well before Cyril Houle began writing on this topic, it was not an area that had received much attention. While the impact of Houle's writings on our understanding of SDL is a matter of debate, there is general consensus that it was noteworthy. Brockett and Donaghy (2011) have noted that "Houle's greatest influence on self-directed learning" (p. 3) was from *The Inquiring Mind*. While many scholars have pointed out that Houle did not use the term 'self-directed learning' in *The Inquiring Mind*, research by Brockett and Donaghy (2011) indicated that may of the field notes and hand-written comments Houle made regarding his research for *The Inquiring Mind* made reference to SDL. Though, we will never know when these notes were made – perhaps they were added after his research was complete – Brockett and Donaghy (2011) suggested that Houle may have considered his research that led to *The Inquiring Mind* a study into SDL. In any event, *The Inquiring Mind* has influenced generations of SDL scholars and has had a substantial impact on the study into adult learning.

Literature related to SDL may now seem like it has reached its pinnacle and is, perhaps, now on the decline; though, this was certainly not the case in the 1960s. Rather, Houle (1961) noted that adult education studies at that time "all pursue a single general theme, the effort to discover the characteristics of those people who are served by one or more of the established institutions or associations" (p. 5). It was apparent to Houle that there was a lack of research into other forms of adult learning outside of formal institutions. This research gap prompted Houle to focus on an area that he believed needed to be explored. More specifically, when referring to the research contained in *The Inquiring Mind*, Houle (1961) noted that "the decision to focus the present inquiry on the individual was reinforced by the perplexing fact that no such studies have previously been undertaken, a gap which has been independently noted by other summarizers of the literature" (p. 9). With that in mind, Houle decided to learn more about continuing learners by having them reflect on their learning and discuss their thoughts and feelings on their learning.

Houle set out to gain a better understanding of why continuing learners feel compelled to seek out learning opportunities by having the learners themselves describe their behaviors. Houle did not choose to study the learning behaviours of "all continuing learners but on those adults who engage to an outstanding degree in activities which are commonly thought to be educational" (Houle, 1961, p. 4). The interviews consisted of 19 major questions and corresponding sub-questions. More specifically, Houle (1961) lists his major areas of inquiry as follows:

These questions were designed to get at the following points: 1. Do continuing learners possess any particular characteristics which make them different from other people? 2. What were the factors that led them to become continuing learners? 3. What has been the history of their continuing education in the past? 4. How much education are they now

undertaking and of what kinds? 5. How do they think society views continuing education? 6. How do they themselves view? (p. 83)

Once the interviews were concluded, "all cases were read and the major themes for analysis were identified" (Houle, 1961, p. 86). The results of this analysis are fascinating to say the least.

Results

Although Houle's study offered insight into continuing learners in a multitude of ways, its most significant contribution to adult learning was the categorization of three subgroups of continuing learners. Houle (1961) described these subgroups as follows:

But while they were basically similar, they did vary in terms of major conceptions they held about the purposes and values of continuing education. As I pondered the cases, considering each one as a whole, it gradually became clear (after many an earlier effort at analysis had led nowhere) that within the group there were in essence three subgroups. The first, as they will be called, the *goal-oriented*, are those who use education as a means of accomplishing fairly clear-cut objectives. The second, the *activity-oriented*, are those who take part because they find in the circumstances of the learning a meaning which has no necessary connection, and often no connection at all, with the content or the announced purposes of the activity. The third, the *learning-oriented*, seek knowledge for its own sake. These are not pure types; the best way to represent them pictorially would be by three circles which overlap at their edges. But the central emphasis of each subgroup is clearly discernible. (pp. 15-16)

The specific characteristics of each subgroup were thoroughly described by Houle (1961) and, from this, it became clear that all continuing learners are not the same.

Learner Subgroups

One subgroup of continuing learners that Houle (1961) identified was goal-oriented individuals. While Houle indicated that the participants of his study were all different, those that were goal-oriented shared many characteristics and/or behaviors. Houle (1961) stated that "the continuing education of the goal-oriented is in episodes, each of which begins with the realization of a need or the identification of an interest" (p. 18). Adding to this, Houle (1961) indicated that "there is no even, steady, continuous flow to the learning of such people, though it is an ever-recurring characteristic of their lives" (p. 18). In other words, goal-oriented continuing learners are always looking for further learning opportunities, with each learning episode having a particular goal.

The activity-oriented continuing learners are markedly different from the goal-oriented learners. Firstly, Houle (1961) stated that "the activity-oriented take part in learning primarily for reasons unrelated to the purposes or content of the activities in which they engage" (p. 19). More specifically, Houle (1961) described activity-oriented people as those who are "course-takers and group-joiners" (p. 23) who

... might stay within a single institution or they might go to a number of different places, but it was social contact that they sought and their selection of any activity was essentially based on the amount and kind of human relationships it would yield. (pp. 23-24)

The activity-oriented continuing learners are keen to learn but would prefer to do so in circumstances that provide social interactions. In fact, this social interaction is of higher importance than the learning itself. This is clearly different from the aforementioned goal-oriented learners and the third sub-group of continuing learners.

The last sub-group that Houle (1961) identified is the learning-oriented continuing learners. Houle (1961) describes learning-oriented individuals as those who cannot stop learning. In particular, for this group, "... the desire for learning may be so strong that it takes on an almost religious meaning" (p. 39). The learning-oriented participants described their desire to learn as pervading all aspects of their lives. In the words of Houle (1961) "for the learning-oriented, education might almost be called a constant rather than a continuing activity" (p. 24). It is almost as if learning-oriented continuing learners have little or no control over their desire to seek out learning opportunities. In short, learning-oriented continuing learners are driven by their desire to learn and enjoy the process.

While Houle (1961) delineates between the three sub-groups of continuing learners, he also makes it clear that all three groups have a number of characteristics in common. This is evident in the following description offered by Houle (1961):

All of the people in the sample are basically similar; they are all continuing learners. They have goals; they enjoy participation; and they like to learn. Their differences are matters of emphasis. Most of them fit clearly into one or another of the three groups but none is completely contained thereby. (p. 29)

An important point to consider is that all continual learners share similar characteristics. While learners in each sub-group have a preferred style of learning and typically approach learning opportunities with different goals in mind, they also all have a strong desire to continually learn.

It is also important to note the limitations of Houle's (1961) findings. Firstly, Houle (1961) interviewed only 20 participants, which limits the generalizability of his findings.

Additionally, Houle (1961) did not explore the reasons why each learner became involved with each individual learning opportunity. While he notes that learners in the three sub-groups share

common characteristics, the specific context in which this is evident is not explored. It is possible that some learning opportunities are designed to encourage the participation of one subgroup over another. This would be less than ideal if universal participation is desired.

Summary

The work conducted by Houle illustrates that individuals seek out learning opportunities for different purposes. As such, understanding why an individual is involved in a particular learning endeavor provides valuable information that can be used to foster an environment that is conducive to learning. Teaching portfolios are a form of SDL that have been used by faculty for various purposes, but there is no research that has described their use in relation to the continuous learner subgroups as described by Houle (1961). Why this is the case is unclear. This study seeks to explore the connection between Houle's (1961) subgroups and the reasons faculty use teaching portfolios. Ultimately, this information can be used to improve the learning environment in which teaching portfolios are prepared and maintained.

Teaching Portfolio Literature

Introduction

Teaching portfolios have been a method of evaluation that is intricately linked with professional development and have been used for several decades in North America (Trevitt & Stocks, 2013). Knapper (1995) noted that "their origins can be traced back to an initiative of the Canadian Association of University Teachers in the early 1970s" (p. 45). In the decades that followed, teaching portfolios became more commonly used by faculty and postsecondary institutions alike. Supporting this, Dinham and Scott (2003) stated that "in education, the use of the portfolio in initial teacher education, in the first years of teaching, in various leadership preparation programs, postgraduate university courses and in teaching awards, has become more

common" (p. 230). Seldin (2000) noted that faculty "are being held accountable, as never before, to provide indisputable evidence of the quality of their classroom instruction" (p. 37). With this focus on accountability and the increasing popularity of using teaching portfolios in postsecondary institutions, it is expected that teaching portfolios will continue to be used in the future.

Teaching portfolios are a form of SDL. Although their use has been studied by scholars, there is still much to learn about their effectiveness as a SDL for enhancing teaching practice. While there is extensive literature describing teaching portfolios, there is limited research exploring faculty perceptions on the experiences of preparing and maintaining a teaching portfolio. In this literature review, I briefly describe the literature related to how teaching portfolios are described, teaching portfolio content, summative and formative uses of teaching portfolios, and whether they are considered effective for these forms of evaluation.

Teaching Portfolio Description

Teaching portfolios have been described in numerous ways. Reece et al. (2001) indicates that "a teaching portfolio serves to document a faculty member's commitment to and accomplishments relative to the scholarship of teaching" (p. 182). A similar description is provided by Weeks (1996) who stated "the teaching portfolio is one way of organizing the collation and presentation of evidence about teaching" (p. 71). Another useful description of teaching portfolios as provided by Wright et al. (1999) is as follows: "we understand a teaching portfolio (or dossier) to be a collection of evidence of good teaching practice, where teaching is seen as everything that faculty do to help students to achieve course and program goals" (p. 90). Oermann (1999) also noted the formative purpose of teaching portfolios by stating that they can be used "to assist in the growth and development of the teacher" (p. 224). While scholars may

describe teaching portfolios differently, it is clear that a key purpose is to present information about many aspects of teaching and serve as a means to improve teaching. However, there is little evidence-based information on how teaching portfolios are used most effectively.

Teaching Portfolio Content

The process of preparing a teaching portfolio can vary, with a few suggested ways to begin the process. Reece et al. (2001) stated that when preparing a teaching portfolio, "the first step is to explicate a teaching philosophy" (p. 182), which is described as "a statement of the faculty member's beliefs, values, and attitudes toward meeting the educational needs of students and toward fulfilling the missions of the department, college, and the university" (p. 182). Another description of a teaching philosophy was offered by Schönwetter, Sokal, Friesen and Taylor (2002): "a teaching philosophy statement is a systematic and critical rationale that focuses on the important components defining effective teaching and learning in a particular discipline and/or institutional context" (p. 84). Corry and Timmins (2009) also indicated that "the first phase of developing a portfolio requires the development of a philosophy of teaching" (p. 389). Lastly, Murray (1995) stated that "perhaps the only essential (at least when portfolios are used for formative purposes) ingredient for a portfolio is some sort of statement of beliefs (a philosophy of education) about the teaching and learning process" (p. 170). There is general consensus that a teaching philosophy statement is an essential component of a teaching portfolio; however, the remaining content can be much more varied.

Although teaching portfolios are unique to each individual faculty member (Oermann, 1999; Reece et al., 2001; Seldin, 2000; Seldin, Miller & Seldin, 2010), there are some guidelines that have been offered to assist in selecting the content. Berk (2005) suggested that there are three categories of components: a) a description of teaching responsibilities, b) a reflective

analysis, and c) evidence to support above claims. Similarly, Corry and Timmins (2009) indicated that a teaching portfolio usually consists of three parts including "an introduction to and description of a personal teaching philosophy. The second outlines and reflects upon entries (discrete teaching experiences) and the final section provides a reflection on the process so far" (p. 390). Likewise, Kaplan (1998) suggested

[T]he portfolio should be more than a simple collection of documents. It also should contain reflective statements on the material included and on the faculty member's approach to teaching and student learning. The reflective portions of the portfolio help set the documents in context for the reader; the materials provide evidence to back up the assertions made in the reflective statement. (pp. 1-2)

While the aforementioned categories of content serve to guide the process of preparing a teaching portfolio, specific components may vary considerably.

A variety of specific components have been used in teaching portfolios, with some used more frequently. Buckridge (2008) notes that at Griffith University in Nathan, Australia

[T]he portfolio is a prose document of about 2000 words which may make reference to a range of Appendices. These consist mainly of evaluation results (student surveys, student comments, peer reviews, etc.), but can include first-order teaching material (subject outlines, tutor manuals, laboratory resource books, etc.). (p. 120)

Another description of teaching portfolio content can be found in Dinham and Scott's (2003) study of teaching award winners in Australia in which the portfolios of 29 teachers were examined. The required teaching portfolio components of the award submissions included a cover page, table of contents, one-page career map, statement of professional beliefs and values, and evidence of professional accomplishment as related to the four broad domains suggested in

the standards framework (curriculum content; pedagogy; assessment and reporting) (Dinham & Scott, 2003).

In addition to the aforementioned examples, Seldin (2000) offers an extensive list of 28 different items that could be included in a teaching portfolio which he grouped into the following categories: 1) material from oneself, 2) material from others, 3) products of good teaching and student learning, and 4) the appendix. Lastly, the Canadian Association of University Teachers (2007) provides an even more exhaustive list of 41 different items that are categorized as follows: 1) teaching responsibilities and practices, 2) products of good teaching, 3) evaluating and improving one's teaching, 4) contributions outside the classroom, 5) information from students, 6) information from colleagues, and 7) information from others.

With such a broad range of content to choose from, it is helpful to note that, depending on the context, the content of a teaching portfolio is adjusted. Dinham and Scott (2003) noted that "teachers also teach in differing contexts – primary/secondary, city/country, art/science – which need to be recognized" (p. 232). Similarly, Urbach (1992) indicated that "different fields and courses lend themselves to different types of documentation" (p. 71). Finally, Seldin, Miller and Seldin (2010) indicated that:

The items chosen for the portfolio are based on a combination of availability of supporting materials, the nature of the portfolio, the faculty position, the discipline, and the importance assigned by the faculty member to different items...differences in portfolio content and organization should be encouraged to the extent that they are allowed by the department and the college or university. (p. 10)

In addition to considering differences in teaching responsibilities, it is important that a teaching portfolio be prepared with intended use in mind. Supporting this point, Murray (1995)

stated that "a portfolio intended to impress a promotion or tenure committee needs to carefully present the best of one's work. A portfolio intended to improve teaching may contain artifacts and reflections on problems one discovers in teaching" (p. 169).

The process of preparing a teaching portfolio can be a new experience for faculty. Seldin (1997) points out is the importance of a mentor who is experienced in teaching portfolio construction to guide the process. This will help to ensure the content of the portfolio is personalized by the instructor preparing it and that all relevant components are included.

With the many demands placed on faculty, it is important to identify how long the process of preparing a teaching portfolio may take. Seldin, Miller and Seldin (2010) claimed that the initial time investment is between ten and eighteen hours that can be spread over a number of days and recommend that the portfolio's length be roughly eight to twelve pages, not including appendix materials. CAUT (2007) guidelines suggest that a teaching portfolio not be longer than three pages. These recommendations differ significantly and may confuse those who are seeking teaching portfolio guidelines to follow.

Faculty may teach hundreds of hours each year on top of the time devoted to other obligations, so the time and energy required to develop a teaching portfolio should be as effective as possible. As such, gaining further evidence-based insights on effective preparation and maintenance of teaching portfolios is needed.

Summative Use of Teaching Portfolios

The content of a teaching portfolio is one aspect that may influence effectiveness. The purpose for which it is prepared and how it will be used are also important. Many postsecondary institutions have chosen to use teaching portfolios as part of a summative evaluation process.

Summative evaluations can be thought of as evaluations that are done at the end of an activity, process, or period of time. The use of teaching portfolios for summative evaluations at postsecondary institutions are many and varied. Buckridge (2008) indicated that "portfolios have been prepared within three different institutional processes – confirmation of tenure, promotion at all levels, and participation in the award course, the Graduate Certificate in Higher Education (GCHE)" (p. 120). The teaching portfolios at Griffith University are, therefore, used as a summary of past success to evaluate whether or not an instructor should be promoted or acknowledged with an award. Dinham and Scott (2003) also noted that "the teaching portfolio performs the function of demonstrating one's capabilities and accomplishments, achieved through documents, artefacts and empirical evidence, judiciously selected and linked together in a theoretically sound and coherent fashion" (p. 231). Again, it appears the summative use of teaching portfolios to assess the past performance of an instructor occurs frequently. This is also identified by Klenowski, Askew and Carnell (2006): "portfolios are found in all phases of education and professional development for learning, assessment, promotion and appraisal" (p. 267). Knapper (1995) also noted the teaching portfolios are commonly used for award nominations – a specific type of summative evaluation. Lastly, Berk (2005) noted that "the teaching portfolio should be reserved primarily for summative decisions to present a comprehensive picture of teaching effectiveness to complement the list of research publications" (p. 57).

It is apparent that many institutions use teaching portfolios for summative evaluation — the results of which are often used for tenure decisions, promotional purposes or for the granting of awards or certificates.

Formative Use of Teaching Portfolios

In addition to their frequent use in summative evaluations, teaching portfolios are also used for formative evaluation. Formative evaluations are concerned with reflection and teaching development rather than assessment for awards or promotions. Through teaching portfolios, previous experience with teaching activities can be used as a catalyst to improve future activities.

The use of teaching portfolios encourages faculty members to use their experiences to help them improve their teaching abilities. Retallick (2000) summed up the process by describing it as learning through reflection. Dinham and Scott (2003) have also noted that "selfreflection plays an integral role in the development of the portfolio" (p. 231). This aspect of selfreflections is also noted by Weeks (1996) who stated that "compiling a portfolio is a process – a process involving research, evaluation and reflection on practice" (p. 72). This instructorcentred ownership of the portfolio is also supported by Dinham and Scott (2003) who suggested that "the portfolio should be considered essentially formative, in that the owner/developer of the portfolio should be looking forward as well as considering the past and the present" (p. 232). Kaplan (1998) noted that due to the self-reflective component, "the portfolio is well-suited to helping faculty examine their goals for teaching and student learning, and compare those goals to the reality of their praxis" (p. 2). The importance of self-reflection on the formative evaluation process stimulated by teaching portfolio use is also supported by Knapper (1995), Murray (1995), Reece et al. (2001), Seldin (2000), and Seldin, Miller, and Seldin (2010). There appears to be widespread support of teaching portfolios as a formative evaluation tool due to – among other things – the reflective component associated with their use.

Teaching portfolios have been described as an effective formative evaluation tool partially as a result of how they are individualized by each faculty member. Supporting this

point, Burns (1999) highlights the importance of faculty ownership of the portfolio: "a teaching portfolio is an organized collection of the professor's work, the content of which varies according to what the professor chooses to include in the collection, and according to whose method of portfolio construction the professor follows" (p. 132).

For teaching portfolios to be an effective component of the formative evaluation process, they need to be reflective, personalized, instructor-driven, and contain both theoretical and practical components. They should be prepared with the intent of improving teaching performance resulting in an enhanced student experience.

Benefits of Teaching Portfolio Use

There are different benefits of using teaching portfolios such as stimulating the improvement of teaching abilities or as a tool to evaluate teaching. More specifically, instructors who use teaching portfolios may reflect on current practices, identify areas in which they could improve, and highlight areas in which they excel. Additionally, teaching portfolios offer a way to showcase their accomplishments in a comprehensive and organized format.

Scholars have indicated that teaching portfolios can be effective for faculty development. When discussing teaching portfolios for developmental use, Buckridge (2008) stated that "without doubt, the portfolio, separated from all pressure to make claims and defend, can be an excellent mechanism" (p. 123). In addition, Dinham and Scott (2003) pointed out the practical benefits of portfolios including "highlighting one's professional strengths and weaknesses and strengthening one's teaching skills" (p. 237). Seldin (2000) also noted "the very process of preparing a portfolio encourages the professor to reconsider teaching activities, rearrange priorities, rethink strategies, and plan for the future" (p. 43). Lastly, Seldin, Miller and Seldin (2010) stated

For most faculty, teaching portfolios actually improve their performance. Experience suggests that if a professor is motivated to improve, knows how to improve, or knows where to go for help, improvement is quite likely. (p. 8)

The perceived benefits aside, Dinham and Scott (2003) also pointed out that through the creation of a teaching portfolio, "there were also more emotional or affective responses to do with satisfaction, value derived, enlightenment, the creation of a lasting 'special' document and inspiring others" (p. 237). In other words, there is a belief the development of a portfolio creates a connection, or a bond, between the instructor and their courses. It is believed this type of connection is essential for faculty to be passionate about their work. Faculty members and postsecondary institutions alike should then see this as beneficial.

Teaching portfolios have also been noted by scholars as a way to assess teaching.

Murray (1995) noted that "teaching portfolios provide professors and chairpersons with a means to assess and improve teaching. Moreover, portfolios provide faculty and chairs with the means to recognize and reward good teaching while still promoting the values of the discipline" (p. 164). This is an important point to consider as it is often difficult for administrators to adequately assess teaching. Supporting this notion, Urbach (1992) stated that "it is unusual for members of faculty committees or administrators, in most universities, to have personal knowledge of your teaching behavior in the college classroom, laboratory, or studio" (p. 73). Addressing this difficulty in assessing teaching, Knapper (1995) stated "the use of dossiers also affords an all-too-rare glimpse into the typically private domain of the classroom, and in this sense portfolios are particularly useful in facilitating peer reviews of teaching" (p. 53).

Teaching portfolios are perceived as a tool that can be used to provide evidence of teaching effectiveness to administrators responsible for evaluations.

Lastly, if the process of creating a portfolio includes an acknowledgement of the institutional objectives, then faculty may be more aware of their role within the broad aims of the institution. Buckridge (2008) observed "there is reason to hope that as staff represent their teaching in this way, their practice will follow suit. In addition, they will increasingly engage (at least rhetorically) with the institution's strategic goals" (p. 122). It is possible, that teaching portfolios may be a benefit to the institution based on the rationale that it is often difficult for an institution to communicate its vision and goals to its faculty. Specifically, teaching portfolios could potentially foster this process.

The perceived benefits of using a teaching portfolio as part of a postsecondary institution's evaluation process are many. Based on the literature, teaching portfolios may have the potential to offer an avenue whereby faculty can critically reflect on their teaching skills and philosophy with the intent to improve teaching effectiveness. Additionally, it is possible that portfolios may allow faculty to develop a connection with the courses they teach, sparking the passion and motivation needed to inspire their students. From an institutional perspective, teaching portfolios may help faculty understand the goals of the institution and align their teaching to help reach those goals.

Alongside the perceived benefits of teaching portfolio use, it is also important to review the literature with respect to the possible concerns of their use.

Concerns of Teaching Portfolio Use

Considering that teaching portfolios can be used for both summative and formative purposes, there are bound to be concerns with their effective use. Buckridge (2008), for example, noted that while developmental portfolios are useful for describing teaching capacity and providing direction for improvement, summative portfolios include what has been done –

indicating areas of success and areas that need to be improved. Murray (1995) also highlighted the potential problems associated with using teaching portfolios for summative and formative evaluation and stated that "if the promise of the portfolio rests in their ability to improve the quality of teaching, administrators would be well advised to avoid using portfolios for summative evaluation" (p. 170). This is an important concern with teaching portfolio use. If they are to be an integral part of continual professional development, one may wonder if they can also be effectively used concurrently for summative evaluation.

Concerns with the nature of the teaching portfolio have also been pointed out by Trevitt and Stocks (2012) who suggested that teaching portfolios may only provide information that assessors want, rather than an authentic description of teaching reflections. This poses the question of whether a teaching portfolio can be authentic if it is used as a summative evaluation tool. Trevitt and Stocks (2012) also noted that authenticity may suffer if faculty preparing a teaching portfolio feel that their teaching values and approaches may be different from those of the assessors. McLean and Bullard (2000) also observed the issue of authenticity as a central concern when judging teaching portfolios. For portfolios to be effective in improving teaching, it is essential that instructors are open, honest, and authentic in their representation of themselves.

Even if there is a process that ensures the authenticity of teaching portfolios, there may still be a significant issue with objectively comparing portfolios from different faculty members. It is difficult to use an evaluation method that requires documentation that is inherently different from one person to the next. This issue of consistency, when comparing and evaluating teaching portfolios, has been addressed in the literature. Trevitt and Stocks (2012) noted, for example, that there is no consistent set of assessment criteria that can be used by assessors; rather, they

have to judge submissions on a case by case basis. Centra (2000) supported this position and noted that "how best to assess the information has not yet been addressed adequately" (p. 87). Another issue with teaching portfolio assessment is that the person who prepares their portfolio may be the expert in the subject areas they teach – therefore it is often difficult to effectively critique their portfolio. Wright et al. (1999) clearly described this problem when he stated that "there is a lack of good research evidence about their impact, partly because the diversity of forms and purposes means that simple judgments about their value are impossible" (p. 92). This diversity poses a very real problem when teaching portfolios are used by institutions to compare the teaching performance of faculty who teach different courses or in different program areas. It is essential to understand that not all teaching portfolios are alike and, by their very nature, will be difficult to compare. From this vantage point, the literature indicates that a framework for teaching portfolio development and evaluation – one that allows for personalization and that an institution can work from –is essential.

A framework is a good starting point but is not without potential problems. A framework would have to align with the institution's goals and objectives and, as such, would influence the content of faculty teaching portfolios. As pointed out by Trevitt and Stocks (2012), this type of pressure may force faculty to meet externally imposed standards which "can lead to the construction of the 'performative self' which is 'a fabricated, socially constructed self'" (p. 250). This is a less than ideal situation if one is concerned with the authenticity of a teaching portfolio – a situation that has been noted in the literature. When Buckridge (2008) analyzed teaching portfolios at Griffith University in Australia, she concluded that "the philosophy of teaching section has tended to standardise around the institution's explicit values, as expressed in generic or 'educational' terms' (p. 122). This theme of formatting teaching portfolios to adhere

to institutional goals was also observed by Dinham and Scott (2003) who highlighted the pitfalls of this adherence: "common consensus is that too much prescription can hinder innovation, reflection and an individualized response and thus any benefit derived" (p. 231). Conversely, Murray (1995) suggested that "it is important for departments to agree on an overall structure of what a portfolio should contain" (p. 169). There is clear disagreement among scholars about how an institution should guide teaching portfolio preparation and assessment.

Lastly, Wright et al. (1999) has noted that faculty may find preparing and maintaining a teaching portfolio to be a burdensome and time-consuming process. If faculty are required to prepare a teaching portfolio and perceive them to be an ineffective use of time, they may resent their preparation as a burdensome activity.

Chapter Summary

Chapter 2 provided a review of SDL and teaching portfolio literature as they relate to this study. The literature that was reviewed provided the foundation from which this study was designed and the data collected. The following chapter explains in detail the methods by which this study was conducted

Chapter Three: Methods

Overview

The previous chapter included literature reviews of SDL and teaching portfolios as they relate to this study. Chapter 2 provided the background on which this research is based and included a history of SDL, various descriptions of SDL, and the ground-breaking research of Houle in 1961. It also reviewed the empirical and theoretical research related to teaching portfolio content, summative and formative uses of teaching portfolios, and possible benefits and concerns of teaching portfolios by postsecondary faculty.

The purpose of this study was to gain insights into the experiences that faculty members in postsecondary education institutions have had preparing and maintaining their self-directed teaching portfolio for the purpose of teaching improvement. This chapter describes the research paradigm guiding the study, the methodology, a description of the participants and how they were recruited, data collection and analysis methods, and, finally, limitations related to the findings of this study.

Research Paradigm

Teaching portfolios are a SDL activity which have been used by faculty for summative and formative teaching evaluations. Although there are some elements that may be common between portfolios, typically they are personalized by each faculty member. As such, the experience of preparation and maintenance of a teaching portfolio will be unique to each participant. As these experiences, and the interpretation of them, are constructed differently by each participant, a constructivist paradigm was used to frame this study.

Mertens (2010) indicated that one of the key assumptions of a constructivist paradigm is a desire of the researcher to understand the world through the experiences of others. The

constructivist paradigm assumes that reality is socially constructed and that individuals will have a different perspective of the world around them based on their own unique experiences (Mertens, 2010). Additionally, Mertens (2010) noted "the researcher's goal is to understand the multiple social constructions of meaning and knowledge" (p. 18). Caelli, Ray, and Mill (2003) suggested that "humans construct knowledge out of their somewhat subjective engagement with objects in their world" (p. 2). In other words, for some problems there are no right or wrong answers – solutions need to reflect the complex reality of each person's experiences. This study explored the experiences of faculty members who have prepared and maintained a teaching portfolio. The results of this study reflect the fact that each participant had different experiences preparing and maintaining a teaching portfolio, which influenced their perceptions of the influence of teaching portfolios.

Methodology

The study was designed to elicit in-depth responses from the participants regarding their experiences preparing and maintaining a teaching portfolio as a self-directed activity for the purpose of teaching improvement. To elicit in-depth responses, the researcher interviewed participants individually, asking them semi-structured questions. As such, generic qualitative research methodology was well-suited for this study.

Generic qualitative research is used to "discover and understand a phenomenon, a process, or the perspectives and worldviews of the people involved" (Merriam, 1998, p. 11). Caelli et al. (2003) described generic qualitative studies as

... those that exhibit some or all of the characteristics of qualitative endeavor but rather than focusing the study through the lens of a known methodology they seek to do one of

two things: either they combine several methodologies or approaches, or claim no particular methodological viewpoint at all. (p. 2)

Additionally, Caelli et al. (2003) noted that in generic qualitative research "the focus of the study is on understanding an experience or an event" (p. 2). The purpose of this study was to provide insight into faculty conceptions of the process of preparing and maintaining a teaching portfolio which fits well with a generic qualitative research methodology.

Generic qualitative research is used when a particular qualitative research methodology does not correspond with a study (Merriam, 1998); yet, it is essential that the researcher follows established processes that are used in other forms of qualitative research. It is the responsibility of the researcher to establish to the reader the merit of a particular study so that it can be evaluated appropriately (Caelli et al., 2003). Mindful of this, this study follows the recommendations of Caelli et al. (2003) for ensuring credibility in generic qualitative research. These guidelines indicate that generic qualitative research address four areas: "1. the theoretical positioning of the researcher; 2. the congruence between methodology and methods; 3. the strategies to establish rigor; and 4. the analytic lens through which the data are examined" (Caelli et al., 2003, p. 5).

Theoretical Positioning

According to Caelli et al. (2003), "theoretical positioning refers to the researcher's motives, presuppositions, and personal history that leads him or her toward, and subsequently shapes, a particular inquiry" (p. 5). Chapter 1 describes key aspects of the researcher's theoretical positioning as it relates to this study. Additionally, it should be noted that the selection of a constructivist research paradigm on which this study is founded is not only relevant to the study itself but also is consistent with the researcher's world view. In advance of

this study, the researcher presupposed that the results of the study would show the experiences of the participants in preparing and maintaining their self-directed teaching portfolio to be similar in some aspects, but also vary, from participant to participant. It was this uncovering of the different experiences of the participants – and perceptions of these experiences – that was of importance. The reasons for this motivation were twofold: 1) to gain insights on how teaching portfolios can be used effectively; 2) to add to the body of knowledge of teaching portfolio preparation and maintenance so others could enhance the use of their teaching portfolio. The methods chosen for this study were selected with these motivations in mind.

Methods

In addition to describing the theoretical positioning of the researcher, Caelli et al. (2003) noted that the methods of generic qualitative research need to be fully explained by the researcher. It is the hope of the researcher that the reader of this study feels fully informed as to the methods used to explore the experiences of faculty while preparing and maintaining their self-directed teaching portfolio. These methods are described in detail throughout this chapter, as are the reasons for why the researcher chose the methods.

Rigor

The approaches that guide generic qualitative research need to be rigorous (Caelli et al., 2003, p. 7). While what constitutes rigor in qualitative research may be debated, Caelli et al. (2003) suggested that "qualitative researchers need to 1) articulate a knowledgeable, theoretically informed choice regarding their approach to rigor, and 2) select an approach that is philosophically and methodologically congruent with their inquiry" (p. 7). The methods used to recruit participants, compile data, interpret data, and present and discuss results have been

explained in detail, are established qualitative research approaches, and are valid for answering the research questions of this study.

The Analytic Lens

Caelli et al. (2003) have suggested the analytic lens includes the "interpretive presuppositions that a researcher brings to bear on his or her data" (p. 8). This study was designed to elicit information about preparing and maintaining a self-directed teaching portfolio. The process of analyzing the data gathered in this study was completed in a number of steps. It was the desire of the researcher to analyze the data by reviewing the interview recordings and transcripts multiple times to minimize the possibility of missing relevant data or misinterpreting data. Additionally, the data analysis was reviewed by the researchers' supervisor to ensure that the interview transcripts and the data analysis were consistent and bias was minimized and assumptions bracketed.

Study Site

The research site for this study was a large research-focused Canadian University.

Prior to participant recruitment, the study was reviewed and approved by a Research Ethics Board at the University of Alberta, including a review of the study recruitment material (see Appendices A & B), participant consent form (see Appendix C), and interview questions (see Appendix D).

Participant Recruitment

Participant recruitment for this study consisted of a few steps. The first step was to compile a list of contact information of university Chairs and Associate Chairs. This information was gathered through the university website. Subsequently, the department Chairs and Associate Chairs were informed of the study via email (see Appendix A) and were asked to support

participant recruitment by distributing a letter of invitation email to faculty members in their department (see Appendix B). Department Chairs and Associate Chairs who agreed to assist in participant recruitment forwarded an email invitation to faculty in their department. Faculty interested in participating were asked to contact the researcher directly for more information or to participate in the study. Lastly, a time and location to complete the interview was agreed upon by the participant and the researcher.

Participants

While participant recruitment was a convenience sample at the researcher's institution, there were some aspects of purposive sampling within the sample selected. Participants were required to be faculty members of the university who have prepared a teaching portfolio. As the purpose of the study was to explore the use of teaching portfolios by faculty, it was necessary that all participants met these two requirements. The use of purposive sampling was chosen by the researcher in order to attract participants for whom the use of teaching portfolios by faculty is likely to happen (Mertens, 2010). The study site was chosen as the recruitment site for a number of reasons: 1) it is a comprehensive postsecondary institution with varied and diverse faculty from which to recruit participants; 2) according to Gravestock (2011), teaching portfolios are an optional source of information that can be provided by faculty for promotion and tenure applications at the institution; 3) the geographic location of the institution was desirable by the researcher who is currently studying at the institution.

Data Collection

The method used to collect data was a single private semi-structured interview (Mertens, 2010) between January 22 and April 24, 2016. The interview consisted of open-ended primary and sub-questions which focused on three areas: 1) participant demographics; 2) teaching

portfolio preparation and maintenance; 3) perceived benefits of teaching portfolio use. Mertens (2010) indicated that interviews are purposeful "when you want to fully understand someone's impressions or experiences" (p. 352). Additionally, Mertens (2010) noted that interviews allow the researcher to "get a full range and depth of information" (p. 352). The purpose of using open-ended questions was to elicit as much information as possible. The use of teaching portfolios is multi-faceted, and it was an objective of the researcher to gather as much data as possible describing the experiences of teaching portfolio use. As a result, semi-structured interviews were the method of data collection selected.

Prior to the interview, interested participants were emailed a participant consent form (see Appendix C) describing the study and their role as a participant. The participants were informed that reading and signing the form indicated they were giving informed consent to participate in the study. Additionally, the participants were informed that they were free to refuse to answer any (or all) interview questions, to quit the interview at any time, or to withdraw from the study altogether if they chose; however, once the data had been analyzed, they could not have their information removed. The participants were asked to sign the consent form and return it to the researcher prior to the interview or to contact the researcher for more information.

The interviews took place in quiet spaces at the University of Alberta and were scheduled at a mutually agreed upon time between the participants and the researcher. The interviews averaged 28 minutes and were recorded using a digital voice recording device for later transcription.

Data Analysis

The participant interviews were transcribed by a third party and then returned to the researcher. Subsequently, the researcher reviewed the transcriptions to ensure accuracy and

revised transcriptions that included errors in spelling or omissions. Once the interview transcriptions were completed, they were returned to the participants for verification of transcription accuracy. The participants were informed that if they would like to make any changes, they needed to do so prior to a pre-determined date; otherwise, the researcher would assume the transcription is accurate.

Interview transcriptions were analyzed using a two-step process described by Mertens (2010). The first step was to prepare the data for analysis. During the first step, the researcher became well-acquainted with the data through the following process: 1) the interview process itself; 2) listening to the interviews a second time prior to transcription; 3) listening to the interviews a third time while verifying the content of the transcriptions. While Leavy and Hess-Biber (2006) suggest that the researcher should do the transcriptions themselves, a third party was selected to transcribe the interviews in the interest of time.

The second step of the analysis process consisted of what Mertens (2010) described as the "data exploration and reduction phase" (p. 425). During this phase, the researcher reviewed the transcriptions and coded the data into similar categories. The categories generally followed the structure of the interview questions themselves, however, there were instances whereby participants provided information while answering one question that pertained to a different question. As a result, the researcher ensured that all data was categorised according to the codes. Additionally, quotes were selected which the researcher identified as being representative of responses in the categories. These quotes were used in the 'Results' and 'Discussion and Implications' chapters to strengthen conclusions presented by the researcher.

Lastly, once the data had been analyzed and coded by the researcher, it was forwarded to his thesis supervisor for review.

Trustworthiness

Trustworthiness in this study was addressed in a few ways. Firstly, the researcher ensured that the transcriptions matched the participant responses in the interviews. Secondly, the researcher provided the participants with the opportunity to clarify or revise any responses from the interviews through the verification of transcription accuracy process. Additionally, the results of the data analysis conducted by the researcher were confirmed by his supervisor. Lastly, the information presented in the 'Results' chapter was presented with numerous quotes illustrating the responses made by the participants rather than rephrasing, paraphrasing, or summarizing in the researcher's own words.

Generalizability

Mertens (2010) stated that generalizability "refers to the researcher's ability to generalize the results from a sample to the population from which it was drawn" (p. 4). Although it would be ideal to be able to generalize results fully across similar populations, this is not possible. The results of qualitative studies cannot be accurately applied to other settings, populations, or other periods of time (Patton, 1990). However, it is reasonable to conclude that if the research methods of a study are sound and credible, then the results will have value with respect to insights that are transferable to other settings.

This study is most transferable to similar populations in similar contexts: postsecondary faculty in a research-intensive university who maintain a self-directed teaching portfolio or who are planning to prepare one, and postsecondary institutions who use teaching portfolios for summative and/or formative evaluations. Mertens (2010) noted that if the researcher provides the reader with a full description of the context in which the research took place, then the reader can decide whether or not the results are transferable to their situation. In line with a generic

qualitative research method, it is the hope of the researcher that this study is presented with "enough detail about the study, the approach, and the methods needs to be included so that the reader can appropriately evaluate the research" (Caelli et al., 2003, p. 4) and to gain insights to other similar situations.

Chapter Summary

This chapter described the research paradigm, methodology, and study site of this study. Chapter 3 also outlined the process by which the participants were selected and how the data were collected and analyzed. Through this aforementioned process, the researcher was able to collect sufficient data to inform the study research questions. Lastly, this chapter discussed objectivity and generalizability concerns related to the results of the study.

Chapter Four: Results

Overview

The purpose of this study was to gain insights into the experiences that faculty members in postsecondary education institutions have had preparing and maintaining their self-directed teaching portfolio for the purpose of teaching improvement. This study was designed to gather information regarding the experiences of postsecondary faculty while preparing and maintaining their self-directed teaching portfolio. Additionally, it was a goal of the research to gain insight into how faculty perceive the effectiveness of teaching portfolios for summative and formative evaluation. The previous three chapters introduced the study and described its goals and significance, presented relevant literature on SDL and teaching portfolios, and, lastly explained the methods used by the researcher in order to answer the research questions.

This chapter presents the data gathered by the researcher through the participant semi-structured interviews. It is organized in the order of the questions that the participants were asked, which were grouped in the following categories: 1) participant demographics; 2) teaching portfolio preparation and maintenance; 3) teaching portfolio effectiveness. The data are presented as a summary of the responses for each question, the individual participant responses for each question as interpreted by the researcher, and quotes which have been selected by the researcher that represent themes of participant responses.

Participant Demographics

There were a total of 21 individuals who participated in this study. Participants were comprised of faculty from a variety of different academic departments including Dentistry, Nursing, Chemical and Materials Engineering, Mechanical Engineering, Civil and Environmental Engineering, Obstetrics and Gynecology, Family Medicine, Psychiatry, Fine Arts

and Humanities, Drama and Theatre, Philosophy, Human Ecology, Earth and Atmospheric Sciences, Physical Education and Recreation, Public Health, and Physical Therapy.

Participant university teaching experience was varied, ranging from 5 years to 37 years. The mean number of years of experience was 23.4 years. Most participants (15 out of 21 participants) had taught at more than one institution with many (11 of 21 participants) having taught at three or more institutions.

A third (or 7) of the participants indicated that they have had formal teacher training. This experience was diverse, ranging from the completion of two teaching-related courses in graduate school to a PhD in education. Two thirds (or 14) of the participants stated that they have had no formal teacher training or preparation; however, almost all of those with no formal training (12 of 14 participants) indicated that they had informal training. There was a wide variety of types of informal teacher training described as workshops, sessions, brown bag lectures, mentorship by senior colleagues, swim instructor experience, courses through extension, peer support, teaching squares, seminars, teaching assistant experience, training courses in a previous career, new professor orientation, festivals of teaching and teaching symposia, faculty development courses, department faculty development, conferences, Centre for Teaching and Learning courses. Only two participants indicated that that they have had no formal or informal teacher training.

A third (or 7) of the participants were currently in a teaching stream; one participant indicated they were in a research stream. One other participant stated they were in both. Most participants indicated their workload consisted of a variety of responsibilities with the most common being a 40/40/20 (research/teaching/service) split. The following were the other workload splits that were indicated by the participants: 40/40/20 (teaching/administration/research), 40/40/20 (research/teaching/leadership), 40/20/20/20

(administration/teaching/research/release for professional practice), 50/50 (research/teaching), 75/25 (research/teaching), (Teaching/research/service), administrative.

As the data indicate, participant demographics were diverse, inclusive of a wide range of departments represented by the participants as well as teaching experience and formal or informal teaching training.

Teaching Portfolio Preparation and Maintenance

The participants were asked a series of questions intended to explore their teaching portfolio preparation and maintenance. While the responses were varied, there were a number of instances when common themes emerged.

The participants responded that they had initially prepared their portfolios ranging from two to 32 years ago. There were a variety of ways in which the participants were introduced to teaching portfolios. A little over half of the participants (11 of 21 participants) first learned of teaching portfolios due to summative evaluation requirements of annual reviews, promotions, tenure applications, job applications, and award applications. The remaining participants first learned of teaching portfolios through colleagues, workshops, brown bag lectures, and various teaching centre events.

The reasons faculty prepared a teaching portfolio were diverse. Two thirds of the participants (or 14) indicated that they prepared a teaching portfolio as a requirement for annual reviews, promotion, tenure, job application, or award nomination. Hence, the initial impetus for the majority of participants to create a teaching portfolio was for summative purposes. As Participant 7A stated, "because I wouldn't have gotten tenure otherwise, so bottom line is it's a requirement, if you don't have it, your application would be denied." Similarly, Participant 5A noted that "it was mostly in connection with the tenure decisions." While the connection

between tenure decisions and teaching portfolio preparation was evident, the participants also indicated that there were other forms of summative evaluation that initiated their teaching portfolio preparation process. For example, participant 2A responded that "so it was really FEC [faculty evaluation committee] driven and faculty requirement driven."

There were a number of participants who stated that they decided to prepare a teaching portfolio for formative reasons. Some participants noted that they prepared their teaching portfolio in order to become a better teacher or to serve as a model to assist their students. For example, one participant (13A) prepared their teaching portfolio "mainly to have a model I guess for my students what a possible portfolio could look like."

In addition to the assortment of reasons stated for preparing a teaching portfolio, there were a wide range of differences in the amount of time the participants spent on preparing their teaching portfolios. The range of time was from roughly two hours to 200 hours. Roughly two thirds of the participants (13 of 21 participants) indicated they spent enough time preparing their teaching portfolios while the remaining participants stated that they could have spent more time preparing theirs.

The responses of the participants unmistakably indicated that the reasons for preparing a teaching portfolio are varied, as is the amount of time it took to do so. Similarly, the guidance and resources available to participants to support this process was varied.

Teaching Portfolio Preparation Guidance and Resources

Faculty use a range of resources and guidance to support the process of preparing a teaching portfolio. In particular, a significant number of participants (9 of 21 participants) noted that they used advice from colleagues. Participant 7A stated

I'm trying to remember back then and I think I probably went to some colleagues who had been through the process. Um, one in particular who was a retiring professor who was very strong on the teaching side of things and I just pretty much asked him what do you think the requirements of a teaching portfolio would look like and he, I think he was the one who would have given me that advice.

Colleagues also supported the initial teaching portfolio preparation process by providing examples from which to follow. Over one third of the participants (8 of 21 participants) indicated that these examples were beneficial. Participant 10A noted that they "looked at many other teaching dossiers or portfolios from colleagues as well, colleagues here but also at other places." Likewise, Participant 17A stated that they "just had to look at the others who did the same thing and to look at their documentation I guess or documents and I just, looked at their format and then prepared mine according to."

Workshops were also an important support mechanism for preparing a teaching portfolio. Eight of the participants indicated that they attended a workshop to assist them with their teaching portfolio. Participant 13A described the workshop attended as follows:

The workshop was helpful, it was pretty well attended, I guess, for some of those workshops, there were maybe 30, 40 people there, so lots of discussion about what it should look like and some examples of what other people had done and I guess some of my other colleagues in my department had gone up from promotion prior to me and so I had seen what their dossier what their whole package looked like for promotion and so I had a multiple sources of support I guess. .

Written resources were also viewed as helpful in preparing a teaching portfolio.

Participant 9A explained the value placed on the written resource:

I don't think I could have done, put it together as well, had I not had that quite detailed pamphlet on here's some things to think about for this section. You know they had a really good outline of common sections to include in a portfolio and common content for each of these different sections. And, I actually use that pamphlet now when I teach the graduate students about creating a teaching portfolio because it's, from my perspective, a great resource.

Participant (19A) described using "a terrific handout that listed all the possible ways you could organize a teaching dossier, so it had, it was three columns wide, about 8 point font and it was a four sided handout, and it was terrific, it listed everything you could possibly imagine."

While the participants indicated that they believed the resources they used were beneficial, they did provide a number of suggestions to make the teaching portfolio preparation process more effective. A number of participants noted the importance of starting their portfolio early in their career. As Participant 2A recommended:

I would institute a professional development component in the faculty that enabled young professors to understand this. And an important aspect of that would be preparation of a teaching dossier, in the context of becoming a professor, being competitive applying for advanced fellowships and faculty positions and then for the promotion and tenure process. Starting early and letting people know what it looks like it can frame up their approach and it makes life a lot easier for them down the road.

The value of preparing a teaching portfolio early in one's career was also explained by Participant 6A:

Knowing ahead of time like when you started on faculty that you needed to do that would have been helpful or knowing about it even as a resident because then you have to

accumulate names of people you've supervised or projects you've supervised or people you've taught and all the evaluations or feedback you don't necessarily still have it, right, so not knowing that until three or four years into being on faculty, then you are scrambling to go back and try and find stuff. So, having a template from the get go and just then a file folder that you can just shove stuff into would have been helpful.

It would appear that the earlier faculty begin preparing a teaching portfolio, the easier the process is and the more beneficial it will be.

Participants also indicated that a well-developed mentorship process would have been beneficial. Participant 9A described this process as follows:

But wouldn't it be nice if you could just have kind of a mentor, that you could check in with once and awhile about it. Like if nobody is really going to look at these, you know other than yourself, it's nice actually to have a mentor and a kind of a checklist of how you are doing, it's like a checkup on how's your portfolio doing and what are you learning from it and you know that kind of thing. I think that could add some motivation and relevance to having a teaching dossier for more than just a personal reason.

This process of interaction with a mentor was also suggested by Participant 20A who stated

[I] needed ongoing assistance in developing it rather than just stay, say here's a how-to session telling you about it and then that relationship ended, I needed, I wanted something more, even like a series of things where, here's some information, go away produce something come back, we'll review that and talk about how to improve.

Participants also recommended other resources that would be of benefit for the teaching portfolio preparation process. Participant 11A suggested "The only thing I would change things like at U of A main campus, they should have some very accessible resources for people especially in

different departments, because I also find through my own research, I've noticed the portfolios can be quite different. Between the disciplines." Support resources should be diverse accounting for the variation in teaching portfolio content and uses. One participant (20A) described in detail their recommendations:

I think that, it would be nice if the university had a method for faculty to build or house their teaching portfolios online, that was consistent, and that provided a framework for people who didn't know what to include so I know that there are documents available to look at, at center for teaching and learning in terms of developing a teaching dossier and what you should include, but if every faculty member had that website space set up and designated and a support person to be able to help them create their teaching statement or their philosophy statement and their, like what do you work on statements, if they had that ability to go and just put it in and it would be housed online for people to be able to update I think it would streamline the annual performance review process, I think it would elevate the scholarship of teaching for faculty across the board and I don't know that it would actually be that big of an investment of time or money on the university's part. So I think that would be a big bang for whatever buck that it'd cost.

While it seems apparent that participants see value in a wide range of support, it may be advantageous for institutions to draw from existing resources rather than creating their own.

Along these lines, Participant 15A stated "I think people need to know what resources are out there in the electronic world that they can use, just to communicate that." Lastly, one participant (14A) recommended that the preparation and maintenance of a portfolio is a partnership between faculty and administration/chairs. Although teaching portfolios can be a very personal document, they should be prepared with comprehensive support and resources.

Teaching Portfolio Content

The participants were asked to list the components of their teaching portfolios. There were considerable differences in the components that the participants included in their teaching portfolios. However, most participants included components that could be placed in the following categories: teaching philosophy, teaching responsibilities, professional development, evaluations, service, and research.

Over two thirds of the participants (16 of 21 participants) indicated that their teaching portfolio included a description of their teaching philosophy. Interestingly, another two participants stated that they had a similar section; however, they had used a different heading. These headings included a "teaching passion statement" or "vision".

The following components identified by the participants were grouped into the 'teaching responsibilities' category: history of teaching, teaching responsibilities, course list, details of specific courses taught (lectures, duration, student demographics), position responsibilities, teaching activities, course outlines/syllabi, sample examinations, sample paper topics assigned to students, clinical teaching (operating room, bedside, classroom formats), undergraduate teaching. post-graduate trainees, teaching other health professionals, examples of lesson plans, examples of specific lectures, examples of exams, graduate student details, teaching list (university courses and teaching outside the university), teaching materials (course syllabi, assignment descriptions, class manuals, examples of assignments), courses taught, course descriptions, examples of student work, examples of unique lesson plans, assessments, sample tests using different methodologies, different teaching methodologies used, links to YouTube lectures, important teaching resources used, something representative from every course taught (i.e., learning objectives).

In addition to the 'teaching responsibilities' category, the participants noted items that were in their teaching portfolios that the researcher has placed in a category labeled 'professional development.' The following items were included in this category: presentations and workshops facilitated (slides included), plenary talks, analysis of teaching, letters of accolades, new course development, course development, course changes, continuing education, how feedback from students is used, processes of development of teaching style, improvements in patient care (clinic coordinator), professional development (conferences or workshops attended), scholarship related to teaching (grants, publications related to teaching and learning), teaching scholarship activity, publications related to education, abstracts that have been presented, faculty development workshops presented, journal reviews done, educational grants received, list of continuing education taken related to teaching (workshops, seminars, conferences).

Another category in which to group teaching portfolio content is 'evaluations.' The items that the participants listed that are included in this category were as follows: teaching information (i.e., end of course student evaluation), student feedback, summative activities, peer review papers, peer reviews/evaluations, student assessments over years (improvements in knowledge), peer observations (letters from senior colleagues who have observed classes), student reviews/evaluations, sample written student evaluations, other components that are included in the annual report, global survey, changes in student satisfaction of a course/courses over the years, student feedback, comments from students, cards from students over the years, peer consultation report.

There are many components that could be included in a teaching portfolio that are categorized as 'service'. These items were as follows: service to the department and profession, service in the faculty and outside (i.e., committees), additional teaching outside the university,

papers about education (distance education, blended learning), teaching contributions (what kind of teaching impact and contributions made beyond the formal classroom) - workshops, scholarship of teaching articles written, workshop evaluations from workshops given.

Another category of items that are sometimes included in a teaching portfolio is 'research.' The participants indicated the following 'research' items included in their teaching portfolios: research philosophy, publication list, grants, research statement.

While many of the components of a teaching portfolio could be placed in the aforementioned 'teaching responsibilities, 'professional development,' 'evaluation,' 'service,' and 'research' categories, there were items that did not fit neatly into any of these categories. Regardless, it is important to note the following items that the participants indicated they included in their teaching portfolios: curriculum vitae, selected lecture notes that have been effective, learning resources that others have used, experiences of courses taught, teaching awards, letters and notes (thank-you notes from students, community partners), honors (invitations to speak at graduation, banquets, awards from professional associations), community outreach (speaking engagements at schools, etc.), miscellaneous (special projects or anything that does not seem to fit anywhere else), teaching awards, thank-you notes from a variety of people, information graphics illustrating research interests and teaching philosophy (flipped classrooms, blended learning, gamification of learning / in e-portfolio), philosophy of mentoring (i.e., interaction with grad and undergrad students), examples of student interaction that reflect philosophies, who I am and what I do, video clip demonstrating what is done in the course (eportfolio), description of what has been done to try and address student feedback to make those courses better, focus and planning for upcoming year, questionnaire for students to assess

teacher, field notes of teacher interaction with students in workplace setting, impact on student development.

Faculty choose which items to include for a variety of reasons. When asked to identify why the participants selected particular components, many of them indicated that they chose the components of their portfolios based on templates or examples they were provided. For example, participant 21A explained that their components were "based on other peoples' examples." Another participant (15A) reflected that their teaching portfolio components came from examples: "probably it came from the literature I'd say." In addition to using the literature or colleagues' examples as resources to guide component selection, workshops were also identified as a resource that was used.

Summative evaluation requirements can dictate which components are included in a faculty teaching portfolio. A number of the participants indicated that they chose the components because they were required as part of a summative evaluation process. Participant 2A noted that "they were components of a formal faculty evaluation." Similarly, Participant 11A responded:

I guess it would have seemed that those are the things that get asked most for in terms of faculty evaluation each year, your annual review on tenure stream and for my tenure documents and for my professorship promotion, all of those things, that's the kind of stuff they're looking for.

While some faculty use an external resource to guide them in their component selection process, others choose components based on personal preference. One particular participant (7A) described the process as being based on personal preferences along with the advice of colleagues. Participant 5A stated that "even a faculty member who didn't know me could look at

the portfolio and make a pretty good judgment about whether I've been a successful teacher." In other words, they were components that would, together, effectively demonstrate abilities as a teacher.

To sum up, the participants identified a varied and extensive list of components that comprised their teaching portfolios. The reasons explained for why such components were selected were numerous. Some components were requirements of summative evaluations, and others were recommended by others, while some faculty choose components for personal reasons.

Teaching Portfolio Components Perceived as Beneficial

There were a wide variety of components of a teaching portfolio that may have a perceived positive impact on teaching. Specifically, almost a third of the participants in this study (6 of 21 participants) stated that the development of a teaching philosophy was beneficial to their teaching. One participant (2A) noted, "writing a teaching philosophy piece is very useful for me." Participant 7A described the process of developing a teaching philosophy as follows:

I think just at the end of the day trying to develop a teaching philosophy. That really was a good self-reflective exercise that made me go back and see what am I trying to do here, what are the standards that I want to live by, what the non-negotiables for me, and so that definitely actually influenced the way I teach and although, I mean, because I have a teaching degree, I mean, I had thought about it, but I had never actually formalized it until that dossier was required at which point in time it was a very good exercise for me to go through because I think it enabled me to focus my efforts and ultimately take those, take those criteria into the classroom. You know in a formal manner, as opposed to just show up and teach type of thing.

While some participants identified what they perceived to be the benefits of developing a teaching philosophy, other participants noted that developing a teaching philosophy was an enjoyable process. Specifically, Participant 4A noted "the fun part is the teaching philosophy. That one I found was the fun to do. Yeah I enjoyed doing."

In addition to writing a teaching philosophy, the reflective component of a teaching portfolio and various forms of student feedback - including written student comments, student evaluations, and course evaluations – were also perceived as beneficial to teaching. When discussing what component was most beneficial, Participant 5A stated, "I find the written student comments most helpful because I feel they speak to me directly and if there was an issue or a problem, I can diagnose it more readily from student comments." Likewise, Participant 9A found the course evaluations particularly helpful and responded that "it would definitely be the course evaluations and the reflections. Most definitely, that's when I get the, I'd say most concrete black and white evidence of my teaching." As with Participant 9A a number of other participants noted the importance of reflection as part of the teaching portfolio preparation and maintenance process. Participant 19A stated that "so part of what was discussed in the teaching portfolio was reflecting on what went well and what went badly at the end of every course." The process of self-reflection was identified by Participant 1A as being beneficial as well and responded that "so that sort of self-reflection. But absolutely doing that self-reflection." Lastly, Participant 16A described specific situations in which reflection has had a beneficial effect on their teaching: "you put yourself in the position of writing down a teaching dossier, why didn't I do, why have I never done a YouTube video or a flipped classroom and so on."

There are many components of a teaching portfolio that faculty perceive as being beneficial for their teaching. The components indicated were diverse, as were the components the participants indicated as being the least helpful.

Teaching Portfolio Components Perceived as Being Least Helpful

There were numerous different components that the participants indicated were least helpful. One participant (7A) noted that "once I have my end of course evaluations I look at them for 20 minutes at the end of the year and I never look at them again so I mean it was strictly for our evaluative people to look at it and go okay this is a decent teacher." Another participant (8A) also expressed that student feedback was non-beneficial, stating "anonymous student feedback. It's useless. Sorry, next to useless, because there are too many anomalies so you can have a high ranking or a low ranking and it has limited or it may not have a lot to do with the quality." Lastly, Participant 15A described teaching evaluations provided by students as follows:

Teaching evaluations, they're all the same. I've gotten really high teaching evaluations, as far as I am concerned they can stop doing them. The [end of course evaluations], forget those for me. I had to do them by policy but they gave me nothing. Whether I get 5 or 4.9 or 4.8, it doesn't really matter, it's the variation of the class, how big it is how small it is, useless.

As these quotes illustrate, participants identified student evaluations of teaching to be beneficial components of their teaching portfolios while others clearly indicated the opposite.

It is worthwhile to highlight that one participant believed that the least beneficial component of their teaching portfolio was their teaching philosophy. This participant (21A) stated that "oh, my philosophy. Total waste. I would never read it when I see other people's philosophy. I actually don't read it; I see people's tenure applications. I don't read it." This is in

stark contrast to the prevailing view in the literature (and some of the responses of the participants in this study) that a teaching philosophy is a valued component of a teaching portfolio.

In addition to student feedback and teaching philosophy, participants noted a variety of other components they perceived as being non-beneficial. These components could be categorized as administrative information that is descriptive but may not lead to improved teaching and included the following: a list of continuing education courses, articles reviewed, grants received, conferences attended, examples of syllabi, course descriptions, committee work, and the numbers of patients treated in a clinic. Participant 2A summarized these activities as I guess sort of the nuts and bolts, it is just an inventory of activities. Yeah, that wasn't helpful for me in my teaching. But it was necessary and certainly let me have an

objective vision of how much teaching I was doing.

Teaching Portfolio Uses

Teaching portfolios are frequently used for summative evaluations. All of the participants indicated that they had used their teaching portfolio for summative evaluation on at least one occasion. Participant 1A noted that "it was more to really provide evidence on why I should be promoted." Likewise, Participant 8A stated that their teaching portfolio was only used "two times, tenure and promotion, that's it." While tenure and promotion were commonly cited teaching portfolio uses by the participants, other uses were also identified. Participant 10A explained that they had used their teaching portfolio "for tenure, for promotion, for awards, for mentoring others certainly in how they do it. Yeah, it's been used in grants. Yeah all kinds of things." Another participant (19A) indicated that they used their teaching portfolio for

"applications, its great source of data for grant applications, so grant applications, awards, and job applications."

A number of participants responded that they had used their teaching portfolios for award and job applications. Participant 5A stated that "I've won some teaching awards and it's been used for those. I've also used parts of it when I've changed jobs, so when I moved to the University of Winnipeg, I had to supply my teaching portfolio before the interview." When reflecting on how they have used their teaching portfolio, Participant 4A indicated it was used for a variety of purposes:

So I'm going to use it for promotion now because I am eligible for promotion and I have used it for awards. It's more for me that I have used it. Honestly. Yeah, but I have to use for our work, for example, if I am even applying for grants, I get information from there. There is a lot of things.

Likewise, Participant 9A described in detail the uses as follows:

Number one use of it has been personal growth. The second use of it though has been to have documentation available to put on annual reports, you know what I accomplished, what my achievements were this year, and so that would be a second thing, because that's a regular thing every year. But I have also used it to pull together a promotion package. I've also used it to put together a teaching award package. So once again, everything is there. I also use it as a teaching tool because I get all of our undergraduate students to do a professional portfolio... What I love about this portfolio is that it does, it tracks everything, it gives me a place to keep everything together and the benefit of that for me has been when I had to compile my package for promotion, wholly smoke, it took me no time at all because everything was in there. It's already compiled, you take things, but

you don't have to scramble through filing cabinets full of stuff. You have all of the relevant information right there and then you just have to pull in current letters from, of support from people, that's all. So, yeah it's just this great central repository of stuff.

While the participants responded that they used their teaching portfolios primarily for summative evaluation, one participant (12A) discussed at length how it was used formatively:

There have been some situations when the actual process of writing out the portfolio, even if it was for some sort of administrative means, curiously enough gave me some ideas as to what I wanted to do for the next year. So, in that sense I would say that while I wasn't forgetting what I was doing, the actual writing of the portfolio gave me time to kind of meditate and contemplate on, "What's the next thing that I want to do with this?" If I hadn't been writing them out, it might not have occurred to me or it might have occurred to me too late to do anything about it where I'd be standing in the classroom the next year thinking, "Oh, I could've done this, but I'm obviously not going to be able to do it now because I'm in mid-semester or something." I think in that sense sometimes just the opportunity to reflect and to be forced to reflect so to speak is not a bad thing.

The participants of this study identified numerous ways in which they have used their teaching portfolios. While not all participants used their portfolios in the same manner, a significant portion used it for summative purposes. Additionally, a number of participants described how they had used their teaching portfolio formatively.

Teaching Portfolio Format

Teaching portfolios may be prepared as a hard copy or online as an e-portfolio. A significant majority of the participants of this study (17 of 21 participants) indicated their teaching portfolio was in hard copy format rather than an e-portfolio. A few participants use e-

portfolios and one participant indicated they have both a hard copy and e-portfolio. Almost two thirds of the participants who currently have a hard copy portfolio (11 of 17 participants with a hard copy teaching portfolio) responded that they would consider creating an e-portfolio. Participant 2A expressed such support of e-portfolios by stating, "I'd be pleased to do that because I think it would be very useful for students who are looking for courses to be able to evaluate the professor and see whether the ideas were modern, interesting, and so on." Another participant (14A) also would consider creating an e-portfolio and described the reasons, stating "I'd favor the public, I'd favor the transparent, I think that's what makes it organic and meaningful so I would welcome that."

There are a variety of resources and support that would assist faculty in creating an eportfolio. One participant (1A) indicated a website template would be helpful:

Well, if the website was already there, I mean if I look at what an e-class website is, but even just if you have the tabs already set up If the tabs were set up and you could put CV goes here, Qualitative Analysis of Teaching goes here, yeah and then just put things in folders that would be great.

A few other participants stated that they would want assistance with formatting in other ways. For example, Participant 2A indicated a willingness to create an e-portfolio "if the service was provided in terms of assistance given somebody to give the file to have it uploaded and some guidance on format." Participant 12A stated that they would want more assistance and responded:

[F]rom a support standpoint, I basically would want to write the thing up and I want somebody else to take care of all of the process of getting it up online and that sort of thing or have it fully automated.

Likewise, Participant 6A would want "someone to do it for me. Make it happen." A couple of participants suggested that a seminar or a workshop would be helpful. Participant 9A described how they would find this useful:

I would need a whole kind of little workshop on how do I go about this. And you know I would really want to, in that workshop, know what the benefits of this would be because I am the age I am, I like hard copy of things. I still like reading books, you know like, actual books. So you know it would, you'd need to really tell me how this would be an advantage to me.

Another participant (18A) described in detail the resources and support that they would want:

So, yeah, if I had a tool to possibly use and some degree of support I think I am reasonably able to use some of that (online e-port sites) myself but at least some degree of support and probably the biggest factor would be time in my schedule to get creative with that, I would love to do that actually...If there was an existing program then again, I can just plug in things or I can add things or, that would be prime, because whether it's created I guess in house or somebody in here whether was it, I don't have the skills to create something like that on the computer I need some existing software type of program that I could use and then it would I guess usually what I find is if I can have somebody at least show me, or give a course that shows how to use it, at least how to initiate, how do you put thing into it, the ground rules of it for the most part I could work with it from there so some degree of training around the software, and at least a resource if I had questions I could go to them and say, how do I do this, or wanted to do this, is that possible. Some kind of help access would probably be sufficient.

E-portfolios are not a desired teaching portfolio format for all faculty. Although most participants who currently do not have an e-portfolio would consider creating one, there were a few who would not. Those participants who indicated they have no desire to create an e-portfolio generally explained that they either didn't want to spend the time on it or they didn't want the information in their portfolios to become public. When asked if they would consider preparing an e-portfolio, Participant 16A stated "no, I only do something if I have to, so if there's an award that in applying for and this is the requirement then yes." Another participant (3A) also expressed concern with e-portfolios:

There's a certain level of privacy that I'd like to maintain to be able to have certain level of distance between myself and the students, so there's a level of professionalism there. I'd put down what classes I've taught. I perhaps talk about how I'm doing teaching and stuff, but this is a lot of extra information, a lot of extra work in doing it and I don't get any benefit for doing it.

Participant 12A explained that they would be afraid that others will use the information that is online as their own (i.e., other faculty members will use the information in their own grant proposals). While Participant 4A also had privacy concerns with a public document:

[I] would like to have an e-portfolio that I can control. If I want it public, I can click and it goes public. Or I want, yes, having the option. To share, yeah or that's another option that I only share with the selected client.

Lastly, Participant 5A suggested that teaching portfolios should not be public and that "those documents, at least to my mind, are for the teacher and for others in the profession who need them for evaluation purposes or who could benefit from their guidance."

In addition to the privacy concerns expressed by some of the participants, two other participants indicated that they would not want to create an e-portfolio for different reasons.

Participant 17A simply said that they did not want to spend time on it, and Participant 21A did not see the value in it.

The responses of the participants indicate that there is mixed support for the preparation of e-portfolios. Additionally, it is clear that a variety of support and resources would be desired if such a task were to be undertaken.

Teaching Portfolio Maintenance

Teaching portfolios are maintained in a variety of ways. The participants of this study responded that they updated their teaching portfolios ranging from never to frequently. However, most of the participants indicated they update their portfolio roughly once per year. Participant 14A stated "I think there's kind of a major look at it once a year and a minor refresher tweak, half way through the year." Similarly, Participant 19A responded that they updated their teaching portfolio "once a year, when I have to do my annual report, unless something comes up that's compelling halfway through the year." Another participant (4A) said they updated their teaching portfolio more frequently – at least every month or two months. Participant 10A also expressed that their portfolio is updated frequently and that items are added or updated throughout the year.

A number of participants responded that they updated their teaching portfolios only when necessary. Participant 3A stated that they did so "during the major events. So, the teaching philosophy, I'll be readdressing as I'm writing my application for promotion but that's about it." Likewise, Participant 2A responded that their teaching portfolio was updated "at least once a year because [faculty evaluation] happens once a year." A similar response was provided by

Participant 12A who stated that their teaching portfolio was updated "minimally probably once a year because I'm doing it for the annual report if for nothing else." A few other participants also noted they only updated their teaching portfolios out of necessity when applying for a new job (Participant 11A), for promotion (Participant 21A), or when needed for an award (Participant 17A).

Teaching portfolios may change when being updated by adding or removing components. Many participants stated that they have added or removed components while updating their portfolios; however, there were differences in what was added or removed. One participant (2A) noted that their portfolio is altered depending on what it is used for. They described these different processes: "the historical portfolio contains everything I've done, but the annual report is usually activity for the year so there are two versions going on. One's incorporating my CV and one's the annual report." Similarly, Participant 5A discussed how their portfolio is adjusted contingent on use:

Yes, depending on what I wanted to use the portfolio for -- if you're being considered for an award, you don't want the Full Monty as it were, which could fill several boxes. So you have a digest version: usually a summary of your scores, a few peer evaluations, and so on.

While a few participants described how they added or removed components depending on what they were using their teaching portfolio for, other participants responded that the content of their portfolio changed as they developed as teachers. Participant 12A described this process in detail:

Yeah, absolutely. That goes in with what I was saying before. Since it's a moving window, it has to be flexible when you're learning new things. I don't think there's ever

been a year when I haven't done something a little bit different. I usually find that there's a couple of sentences on my teaching philosophy that I always keep because they're really at the core there and I believe that. But, the rest of it changes quite a bit.

A similar reflection was offered by Participant 14A who responded:

A teaching portfolio for me now is extremely different in terms of what's in there because of when I've stopped teaching and what I'm doing now, I am doing way more mentoring and I'm doing way more faculty development, and I'm doing way less direct undergraduate teaching like the big classes and less direct resident teaching so that's obviously going to be reflected in what I have to say in my teaching portfolio.

There were some participants who stated the components of their teaching portfolios have stayed the same; however, many indicated the content within some components has been changed. Participant 11A illustrated this point, stating "not components. Well, I've changed sections, like removed or added what's in a section for sure." Likewise, Participant 18A noted "I haven't added or changed the components they've gotten bigger with more information."

The process of maintaining a teaching portfolio is unique to each faculty member. While some have left their teaching portfolios virtually untouched since they were first prepared, others are continually updating content and have removed or added components over the years.

Teaching Portfolio Encouragement

Encouragement to prepare and maintain a teaching portfolio varies from department to department. The majority of participants (15 of 21 participants) indicated that faculty members in their department are encouraged to prepare or maintain a teaching portfolio. However, among these participants, there was variation as to how this encouragement occurs. A number of participants noted that teaching portfolio encouragement in their department is intricately linked

to the faculty evaluation process. Participant 2A noted "they are formally encouraged to do that because it is enshrined as part of the Faculty Medicine in Dentistry, Faculty Evaluation Committee process." A similar response was provided by Participant 5A: "we use that as the basis for comment on whether the candidate should be tenured or promoted." Likewise, Participant 7A described how teaching portfolios in their department are encouraged as part of summative evaluation processes and stated that they are encouraged "only for the purpose of promotion and if they went out for a teaching award. Otherwise, no they would not be encouraged."

Although many participants stated that there was no departmental encouragement to prepare a teaching portfolio, a few indicated that components of a teaching portfolio were encouraged. Interestingly, one participant (9A) stated that the faculty in their department "are encouraged but because it's not required, there's not much incentive." It would seem that participants of this study work in departments where teaching portfolios are more valued as a component of summative evaluation over formative.

Teaching Portfolio Effectiveness

Teaching portfolios have a varied level of influence on teaching. The process of reflection that is a part of preparing and maintaining a teaching portfolio was identified by some of the participants as being a catalyst for change. Participant 1A noted that over time their teaching portfolio helped them reflect on their teaching, which was perceived as making the course better. Participant 12A responded similarly, stating "as I said, each year that I'm doing this, the process of writing out the portfolio has been a good time for me to reflect on, Okay, what worked and what didn't?" Another participant (9A) explained how this process of reflection "has been this on-going commitment to the reflecting, to the looking at feedback

constructively, because if you've got to write reflection about something, you have to be contextualizing it and be constructive about it." It appears that for some of the participants the process of reflection associated with teaching portfolios was perceived as being beneficial.

The process of developing a teaching philosophy was also indicated by some of the participants as having an effect on their teaching. One participant (7A) acknowledged the impact of preparing a teaching philosophy: "developing a teaching philosophy that reminded me of what I wanted to do and why I was wanting to do it." Another participant (10A) reflected on how developing a teaching philosophy influenced teaching practice:

Actually having to work on, regularly, on your teaching philosophy, changes the emphasis of what you do in the classroom. I believe that you can't, you can't examine what you're doing in the classroom without it having an effect, hopefully.

The effect of developing a teaching philosophy was also described by Participant 20A who indicated that it was helpful in ensuring that the actions as a teacher are in line with the teaching philosophy. For many participants, it would appear, the process of writing and developing a teaching philosophy resulted in a perceived positive effect on their teaching.

Lastly, one participant (4A) specifically identified areas of teaching that their teaching portfolio has affected. This participant reflected that it helped improve teaching by being a stimulus to use more technology, videos, podcasts, and to be more interactive.

While many (15 of 21 participants) of the participants perceived their teaching portfolios as having a beneficial effect on their teaching, this view was not consistent among all participants. A few (2 of 21 participants) participants shared that they were unsure as to the effect that teaching portfolios had. Participant 8A indicated that some components (data gathering part, peer to peer discussion) lead to improvements but this information doesn't

necessarily need to be in a teaching portfolio. It was stated further that effective teaching may come naturally to some people, eliminating the need for a teaching portfolio. Additionally, other faculty members may just naturally try to continually improve their teaching and, likewise, do not need a teaching portfolio to do so. Another participant (13A) noted that the process of initially preparing a teaching portfolio helped but currently there are other preferred avenues for professional development. Lastly, Participant 16A noted that it is difficult to measure teaching improvement and what influences teaching improvement.

A few participants responded that their teaching portfolio did not influence their teaching. One participant (21A), when asked if their teaching portfolio affected their teaching, stated "No. It's quite clear, I just use this to develop my portfolio because it's required for my tenure application so that was the beginning and the end of it." Another participant (17A) also concluded that their teaching portfolio did not affect their teaching, stating "teaching effectiveness has to come from within the teacher and is not the result of a teaching portfolio. It may have a positive effect on some people but it's hard to say." Lastly, Participant 6A stated that "I would say probably not. Hmm, because I don't do it very regularly, right."

The influence that preparing and maintaining a teaching portfolio has on teaching is ambiguous. While some faculty believe that their teaching portfolio has had a positive effect on their teaching, others do not. There is also doubt as to the ability to measure changes in teaching and/or to determine the causes of those changes.

Do Teaching Portfolios Improve Teaching?

Teaching portfolios are used for different reasons. Some faculty may not value them personally while still recognizing that they may be beneficial for others. The participants of this study were asked if they believed that teaching portfolios improve teaching. This is different

from another question that asked if teaching portfolios influenced *their* teaching. While some (9 of 21 participants) of the participants believed teaching portfolios improved teaching, many of them (10 of 21 participants) were unsure if they did.

The participants who indicated teaching portfolios improved teaching stated so for different reasons. The value of the reflective component of teaching portfolios was noted by many participants. Participant 1A believed they improved teaching "because of that reflective component." This sentiment was also described by another participant (5A) who noted that "anytime we get a chance in our busy lives to reflect on what we do, we benefit from it and that's true for teaching as well." One participant (20A) indicated that continual reflection on teaching and using that reflection to make changes to teaching is very helpful. It was also noted that it helps keep faculty accountable to themselves and their students.

The structure of a teaching portfolio can also have a positive effect on teaching.

Participant 4A responded by stating "they structure you. It's like a guideline for being a good teacher." One participant (18A) described this guiding process in detail:

If you're already doing all that stuff, then you're just documenting it or putting it in this place where you can record it, but I think for a lot of our people, particularly as they get into this and start doing that, it actually helps them realize that there are components that are missing. In other words, maybe the scholarship side of the teaching, maybe engaging in mentorship, there is a whole bunch of other areas around teaching that I think a portfolio helps you realize – that if you really want to be considered as a well-rounded teacher and making contributions to education those are the areas you should be looked at.

The development of a teaching philosophy was also identified as having a positive effect on teaching. Participant 10A suggested that "working on your teaching philosophy and examining what you're doing in the classroom will have a positive effect on your teaching."

While many of the participants (9 of 21 participants) indicated that they believed teaching portfolios improved teaching, others were not convinced. Problems associated with measuring teaching improvement and what influences teaching improvement were identified as possible difficulties in assessing the value of a teaching portfolio (Participant 16A).

It was also noted that teaching portfolios may only improve teaching if the reasons for their preparation are intended for such a purpose. Participant 12A explained:

Assuming that the person writing them wants that. If your only purpose of doing it is, "I need to write a paragraph every year just to keep the administrators happy," if you aren't going through some sort of reflective process or thinking about what it is that you're doing, if you don't have a teaching philosophy so to speak, then it probably doesn't do that much good.

A similar response was offered by Participant 15A who also suggested that their use may determine their value. If it is being used to improve teaching, then it can be a good tool; if it is just for a teaching award or tenure, then it is not. Hence, if faculty are forced to create a teaching portfolio for administrative purposes, it may devalue their formative effects.

Teaching portfolios may not, in themselves, improve teaching; however, they may be a catalyst for improvement by initiating a process that may improve teaching. This effect was described by Participant 11A:

I think conscious pedagogy and research improves teaching, and I think a lot of teachers actually don't do that at all. So, it is possible that through doing a portfolio they might be encouraged to do that, but for me pedagogy and research has improved my teaching.

Teaching portfolios may encourage faculty to evaluate their teaching and, as such, stimulate change.

Faculty motivation may also have an effect on the value of teaching portfolios. One participant (3A) noted some professors only do the minimum to improve their teaching because they are more concerned about their research (and this view is reinforced by senior management). However, this participant (3A) noted further that others take their teaching more seriously and will benefit from a teaching portfolio. Similarly, Participant 9A highlighted the effect of competing priorities and stated that "there will be people who, like you were saying, I don't have time for this because I've got so much research work to do." The potential for teaching portfolios to improve teaching is linked with the environment within which they are used.

Two participants responded that they did not think they improved their teaching.

Participant 8A noted that preparing and maintaining a teaching portfolio is not important when one is aware of one's own strengths and weaknesses. This participant continued by stating that teaching portfolios take time that one might not have, or time that is better spent elsewhere (8A).

The perceived ability of teaching portfolios to improve teaching is mixed. Although some study participants believed teaching portfolios improve teaching, others were not convinced. It is difficult to determine with any certainty the exact effect teaching portfolios have on teaching.

Teaching Portfolios for Formative Evaluation

Support for teaching portfolios as a tool for formative evaluation is variable. The majority of the participants (16 out of 21 participants) indicated that they think teaching portfolios are effective for formative evaluation; however, there were a wide range of reasons provided for this belief. The reflective aspect of preparing and maintaining a teaching portfolio may be particularly beneficial in influencing perceived changes in teaching. Participant 18A noted the formative effects of a teaching portfolio "particularly if they got into that reflective piece as an instructor...that to me is really about that cycle of improving teaching, right?" Participant 13A also described in detail how reflection has influenced how they have used a teaching portfolio:

I think as you go through the process of developing portfolio, if it takes you a few week or a few months and you spend some time on it before maybe you have a class that you're teaching a term, and you're developing that over the term, thinking about that portfolio, thinking about and reflecting on your teaching in the term that you're teaching maybe you're going improve things or try new things within that term because you're thinking about how your teaching has gone and what feedback you've received, so I think it would be useful for both for and advance of the feedback.

Teaching portfolios may be effective for formative evaluation but only in particular circumstances. Participant 19A reflected:

It depends on how you do it, the way that I've tackled it has been more about data archiving than reflection so that's a different take on the story, so my particular version is not designed for formative feedback, so no. But if you've included the reflections it would absolutely be a part of you formative assessment feedback, so it really hinges on the pieces that you decide to include.

The formative value of a teaching portfolio may hinge on how faculty integrate them in their teaching development. As Participant 12A stated, "there's nothing that I'm writing in the statement that I didn't know already. It was more that I had to integrate it a little bit and take time to think about it."

Teaching portfolios may provide structure for faculty to reflect on their teaching.

Participant 5A noted that "a teaching portfolio forces even those who teach a lot to monitor the job they're doing in the classroom." Another participant (10A) described teaching portfolios as a framework for faculty to explain what they do, why they do it, and how they do it. Participant 10A continued by suggesting that teaching portfolios also provide flexibility for faculty from different departments. In other words, teaching portfolios facilitate SDL by providing a framework for evaluating their teaching.

Support for teaching portfolios as an effective tool for formative evaluations was not unanimous among the study participants. The limited use of student evaluations for evaluating teaching was noted by Participant 7A: "[the] only thing that affects me formatively is every year I look at my [student evaluations] and I go ok am I doing a decent job with my teaching and you know we move on." Participant 8A echoed this sentiment:

It's really not important. What's important is that in the workflow you think about the work that you do and try to do better every day, and you know if you can gather data that can help you to do that. Right, and that is not the same as preparing a document because it is something you do innately or sometimes it's an innate thing and I don't think preparing documents about it helps.

Lastly, Participant 21A offered that teaching portfolios may be effective as a formative evaluation tool, but they were not for them.

It would appear that teaching portfolios may be effective for formative evaluations, but this is not a perception held by all faculty. Additionally, the reasons faculty perceive teaching portfolios as effective or not are varied.

Teaching Portfolios for Summative Evaluation

Teaching portfolios are perceived as effective for summative evaluation by most participants of this study (17 of 21 participants). A teaching portfolio provides a framework to explain teaching practice, as well as why and how it is being done. Participant 4A stated "it provides all of the evidence" for evaluation and Participant 14A noted it provides transparency of teaching practices. This aspect of providing evidence of teaching effectiveness was explained as follows:

I would think that a portfolio would be a way you can demonstrate teaching effectiveness or certainly the process you have gone through in your teaching to look at teaching improvement so I think it is a component of evaluation for teaching for sure as a summative component. (Participant 13A)

Another participant (5A) described how teaching portfolios can be effectively used for tenure and promotion decisions:

Well, I think they are very useful for tenure decisions. I've never felt I could in good conscience support someone for tenure or promotion whose teaching is weak, or who hasn't provided evidence of being a good teacher (or else that they are working hard to address any problems that have been identified).

The practicality of using a teaching portfolio for summative evaluations was also identified by a few participants (2 of 21 participants). Participant 2A explained "it's a place where you can incorporate that information, integrate it, and gather it together. Without a

teaching portfolio, the information is scattered around and there is no incentive to put it together." Teaching portfolios can be a "great repository it's quick and easy to pull that information up and put it into the reviews, applications, promotions etcetera" (Participant 18A). As such, faculty members may find teaching portfolios useful in organizing information that could be used for a variety of summative evaluations.

Two participants indicated they were unsure if teaching portfolios were beneficial for summative evaluations and others said they may be beneficial depending on how the institution uses them. The validity of teaching portfolios as a summative evaluation tool was questioned due to the accuracy of their content. A concern was that "people can write anything down on paper and it doesn't actually mean anything" (Participant 11A). It is possible that the truthfulness of a teaching dossier's content could be questioned.

The process of evaluating teaching dossiers – particularly across disciplines – could be problematic (Participant 12A). Additionally, participant 12A observed the amount of time that an institution takes to review them, rather than focusing on student evaluations, is important. The focus of research-intensive universities where research and publications are, in some contexts, considered more important than teaching may limit the amount of time and resources an institution dedicates to teaching portfolio evaluations (Participant 3A). As Participant 7A expressed, "I really do question the validity of it, because I don't think it's particularly valued, which is sad, but that's the way I look at it." It is clear that for some participants, it is necessary that teaching portfolios are valued by an institution for them to be seen as effective for summative evaluation.

The difficulty with assessing a teaching portfolio was noted by some participants (3 of 21 participants). Participant 3A suggested that "it would actually have to be put into a mode that

you can actually compare different people and that's a very difficult thing to do." Additionally, evaluators need to be trained to understand the differences inherent in a personal document such as a teaching portfolio. As Participant 9A describes,

I think you have to have a trained eye to look at teaching dossier for summative purposes, you need that trained eye to look at what are the indicators of growth, what are the indicators that you are committed to teaching and are getting better at it.

The importance of training teaching portfolio evaluators was best described by Participant 12A:

I'm not sure if the average administrator would know how to read them. I'm not entirely sure...there's a little bit of an element there for it to be effective, it has to be effective in the sense that the administration can actually evaluate it properly. I'm honestly not sure who's actually doing that or how good they would be at it...I think there's the potential for it. But, I think the other problem is – and I'm sure this is a major issue when you're talking about evaluating it for professional means – is that you're talking about comparing people across a whole bunch of disciplines and different classes and different class sizes. I know that if I was sitting on the other side of the table, I would really want something quantitative. That doesn't mean that I'm going to completely buy into the teaching evaluations because there's obviously issues with that. But, it's going to have to be, if you're looking at people's teaching statements, it's going to be incredibly subjective. If somebody told me, "The difference between you getting an extra half increment is going to be based on your teaching statement," I'd be bothered by that...So, I think in theory it could have tremendous potential, but I'm not sure in practice if it does.

Lastly, one participant (8A) noted that teaching portfolios are beneficial for more significant summative evaluations like tenure and that annual evaluations currently don't need

the quantity of information that is in a teaching portfolio. Perhaps faculty would find value in guidance as to what to include in their teaching portfolios so that they do not get burdened by creating one that includes content that will not be used for evaluation.

Are Teaching Portfolios an Effective Use of Time?

There is varying agreement among faculty that preparing and maintaining a teaching portfolio is an effective use of time. Two thirds (14 of 21 participants) of study participants stated that they believed it was an effective use of time, with a variety of reasons given for such responses. For example, some participants believed that they were an effective use of time because of how they can be used to guide teaching. To this point Participant, 4A stated "Oh yeah. Yeah of course, it will give you so much guidance and so much organization that you really need it." Additionally, a few of participants (2 of 21 participants) responded they were an effective use of time due to their administrative benefits. Participant 19A suggested they were an effective use of time as it helps with filing and organizing information. A similar sentiment was provided by Participant 10A:

Yes, because you are going to need it along the way and if you don't prepare it as you go along, it's going to take you longer to do it, when you have to have it...So, it's much easier and so much more effective and beneficial if you are working on it regularly.

It was noted by one participant (12A) that the amount of time devoted to their teaching portfolio was not excessive and that "I guess I'd say yes because I don't feel like I ever really had to spend a lot of time on it."

Although many participants (14 of 21 participants) responded that they thought teaching portfolios were an effective use of time, others offered different opinions. A number of participants indicated that some aspects of the teaching portfolio were an effective use of time

but not all. Participant 18A believed the reflection part is a valuable use of time, but other components (list of courses taught, student evaluations, conferences attended, courses taken) were not as valuable as they are only used for summative evaluation.

A few participants suggested that teaching portfolios are an effective use of time if used in the right context. One participant (14A) believed that they are an effective use of time as long as the person who writes it and the person reviewing it work in partnership. Participant 6A noted the importance of support: "it would be nice, like I said, if you had a bit more support of good ways to prepare or good ways to do it and then I think it would be helpful for us to maintain." Likewise, Participant 2A was hesitant to state they were an effective use of time and noted "it can be. I think it depends on how much guidance or how to the extent to which one understands what a teaching portfolio should do." Lastly, one participant (17A) suggested it is a more effective use of time when it is done properly over time rather than one point in time. To these particular participants, teaching portfolios are an effective use of time only if they are used in the right environment.

While most participants (14 of 21 participants) indicated they thought teaching portfolios were an effective use of time, a notable number (6 of 21 participants) of participants did not believe so. One participant (3A) believed that in the current state of the University, teaching portfolios were not an effective use of time. This participant explained that research is valued more than teaching and only relevant when there is a problem with one's teaching. However, this participant also explained that teaching has an impact on thousands of students and should be valued more. Two participants (7A and 8A) responded that they were not an effective use of time and they were necessary only for tenure or promotion. Lastly, another participant (13A)

responded that they were an effective use of time, but offered the following explanation for this response:

For me, personally probably not, given the other avenues that I have for reflection and discussion with my colleagues that are probably more valuable in a group setting, there certainly is value in reflection and thinking about that on my own, but whether the time invested to continually sort of go back to that and think about it maybe for some people it would be a good investment for them, and for me I think I would prefer the other avenues that I have.

There were no consistent responses by the participants when asked if teaching portfolios were an effective use of time. While most (14 of 21 participants) participants indicated that they believed they were, many thought they were only an effective use of time in the certain contexts, and a few (6 of 21 participants) participants stated they were not an effective use of time.

Would You Recommend a Teaching Portfolio to a Colleague?

When asked if they would recommend preparing and maintaining a teaching portfolio to a colleague who does not have one, the majority (15 of 21 participants) of participants indicated that they would. The reasons given for these responses were varied. A few (4 of 21 participants) participants explained that they would recommend them as they can serve to help them get a job or get promoted. Participant 5A stated "in my profession, for sure, because I don't know of any institution that wouldn't consider teaching as a major criterion, not just for tenure and promotion, but for being hired in the first place." Two other participants (7A and 15A) would recommend them as they help in tenure preparation.

In addition to recommending teaching portfolios for summative purposes, the participants provided a number of reasons they recommend them for formative purposes. Participant 3A describes this as follows:

My research will have impact to a small community maybe, but my bigger impact will be for the people of the undergraduates, teaching engineers that go out to work. While doing that teaching portfolio, developing a philosophy, developing mechanisms for assessing your own performance and continuing to develop that, will allow you to influence how you teach and how well you teach and the performance of those people in their careers.

One participant (13A) would recommend teaching portfolios for their summative and formative benefits. This was described as follows:

Well, I would think if that person does not have sort of access to the teaching networks or resources and groups that are available in some departments or some institutions, and maybe you're a newer teacher and you're looking for ways to improve and one of the things you would need to think about to improve and to demonstrate your teaching abilities I would think it's a worthwhile investment and I think certainly for anybody that is trying to demonstrate to somebody else what you do as a teacher and processes that you've gone through, I think that is valuable to have a portfolio.

Some participants gave other reasons for recommending a teaching portfolio to a colleague. Participant 2A described how a teaching portfolio can improve the student-teacher relationship and can assist in mentoring students. This was explained as follows:

If you have a reputation as a good professor who is interesting, accessible, and has a dynamic interactive style, then students seek you out. And some of those students are the brightest and best and they want to come and do graduate studies or undergraduate

projects. And that just helps the whole thing move forward, so it increases the capacity for mentorship.

Two participants (6A and 18A) explained that they would recommend preparing a teaching portfolio as they can be helpful as a simple repository of critical information. Finally, one participant (9A) responded that they would recommend teaching portfolios and added "I think I would like to see everybody have them."

Although most participants responded that they would recommend preparing a teaching portfolio to a colleague who does not have one, there were some alternative responses. One participant responded that they would only recommend it if it was necessary for tenure, and a few others stated they would recommend one only in particular circumstances. Participant 12A described who they would recommend teaching portfolios to:

I would say for somebody who's junior, I would strongly recommend it. But, if you're a senior professor, I don't think you're going to change as an educator. You're either already a good teacher or you're not. Chances are if you're not a good teacher, it's probably because you don't care. Having one at that point I don't think would make any difference. In that sense I would say it probably depends on career stage.

Similarly, participant 16A responded that they would not recommend teaching portfolios to all colleagues but would for Ph.D. students. They explained this position as follows:

Well I basically don't tell my colleagues what to do but if you rephrase that as like a Ph.D. student then I'd say yes because they'll need it for their job interviews, they need to think about their teaching and they'll need it to maybe even guide their, you know, their teaching, so yeah for them it's definitely a good idea.

Lastly, one participant (8A) responded that they would not recommend preparing a teaching portfolio to colleagues unless they are coming up for tenure and gave this reasoning for their response: "there is way too much stuff to do and I would say that would fly in the face of the new strategic plan for the University."

It appears that faculty would, for the most part, recommend preparing and maintaining a teaching portfolio to a colleague who does not have one. However, the context in which this recommendation would occur varied among participants.

Chapter Summary

In this chapter, the results from the study were presented, beginning with participant demographics, continuing with interview data related to teaching portfolio preparation and maintenance, and then ending with teaching portfolio effectiveness. The results were presented as to paint a picture of the varied experiences faculty have when preparing and maintaining their self-directed teaching portfolio.

The results answered the research questions presented in chapter 1 by providing insight into the experiences faculty members have had preparing and maintaining a teaching portfolio. It is clear that teaching portfolios are prepared in many ways, have varied content, are used for a variety of purposes, and are valued to varying degrees as a tool for summative and formative evaluations.

In the following chapter, the results of this study will be discussed in relation to the literature review in chapter 2 as well as the implications of these results.

Chapter Five: Discussion and Implications

Overview

The previous chapters introduced the study and presented key aspects of this research.

There was also thorough discussion of relevant literature on faculty teaching portfolio use and SDL. Various components of the research methods used in this study were explained and, finally, the results of this study were presented at length. This chapter discusses these results and identifies some implications.

The purpose of this qualitative research project was to gain greater insights into the experiences that faculty members in postsecondary education institutions have had preparing and maintaining their self-directed teaching portfolio for the purpose of teaching improvement. The participants were asked a series of questions that explored the process of preparing and maintaining a teaching portfolio as well as the perceived influence that teaching portfolios have on teaching. This chapter discusses the results of the study and presents some implications of these results. This study has reinforced some of the literature on teaching portfolio uses, content, and effectiveness while also providing new insight into self-directed teaching portfolio preparation and maintenance. As a result, the findings of this study can be used to inform faculty teaching portfolios as a SDL activity.

Teaching Portfolios as a Self-Directed Learning Activity

The importance of SDL as a topic for study in adult education has been noted by many scholars (Brockett & Donaghy, 2011; Brockett, Stockdale, Fogerson, Cox, Canipe, Chuprina, & Chadwell, 2000; Caffarella, 1993; Garrison, 1997; Guglielmino, Long, & Hiemstra, 2004; Mezirow, 1985). It has been suggested that SDL can occur in a wide range of situations, including "self-planned and self-conducted learning projects" (Guiglielmino, Long, & Hiemstra,

2004, p. 1) and has been described as a study form in which an individual has control over planning, implementing and evaluating the effort (Hiemstra, 1994). This study supports the assertion that teaching portfolios are a SDL activity.

Numerous academics have indicated that individual control of the learning process is a key feature of SDL (Brookfield 1985; Garrison, 1997; Hiemstra, 1994; Knowles, 1975). The participants in this study often noted that they used numerous sources of support when preparing and maintaining their teaching portfolios. While participants indicated that they used guidance from colleagues, attended workshops, followed the examples of others, or used other resources or sources of support when preparing and maintaining their teaching portfolio, they did so as they desired. One such example of seeking out support from colleagues was provided by participant 7A who stated that "I think I probably went to some colleagues who had been through the process was". Similarly, another participant (13A) described a workshop which offered a variety of resources and teaching portfolio examples from which they could draw from when preparing their own teaching portfolio. This aspect of SDL whereby "learners move in and out of learning networks and consult a range of peers" (Brookfield, 2009, p. 2615) appears to be an essential component of teaching portfolio preparation and development. Additionally, participants clearly showed patterns of "diagnosing their learning needs, formulating learning goals, identifying human and material resources for learning, choosing and implementing appropriate learning strategies, and evaluating learning outcomes" which have all been noted by Knowles (1975, p. 18) as key aspects of SDL. This is evidenced in the wide range of instances in which participant teaching portfolios were updated. Participants indicated that their teaching portfolios were revised depending on their particular needs – sometimes regularly, at times sporadically, or for specific uses.

In addition to the importance of support from colleagues and other resources noted by study participants, so too was the freedom to decide what would be the final product of their learning experience. The participants regularly indicated that the format, components, and specific content of their teaching portfolios were, in most part, decided by themselves.

Participant 7A indicated the content of the teaching portfolio was based on personal preferences.

Another participant (5A) noted that the content of their teaching portfolio was based on a desire to demonstrate successful teaching. Additionally, participants also indicated many ways in which they updated their teaching portfolio by removing, adding, or revising components – a process that varied between participants. The participants regularly indicated that the format, components, and specific content of their teaching portfolios were, in most part, decided by them. This study provides ample evidence that the preparation and maintenance of a teaching portfolio by faculty is a SDL exercise, and it also sheds light on when this is most likely to elicit the most effective results.

The results of this study indicate that teaching portfolios can be an effective form of SDL. The data also indicate that there are conditions whereby this is more likely to occur. The participants of this study indicated that various sources of support and resources were essential to the preparation and maintenance of their self-directed teaching portfolios. As such, it would appear that there should be a wide range of support mechanisms in place for faculty who are preparing a self-directed teaching portfolio. Faculty should be made aware of these resources and should have the ability to use the ones that they perceive to be most beneficial.

Teaching Portfolios and Self-Directed Learners

The research of Houle in 1961 has been considered particularly influential on the study of self-directed learners by many scholars (Brockett & Donaghy, 2011; Caffarella, 1993;

Confessore & Confessore, 1992b; Guglielmino, Long, & Hiemstra, 2004; Hiemstra, 1994). In particular, Houle's (1961) research resulted in the categorization of self-directed learners into three categories when describing their reasons for participating in learning activities: (a) goaloriented, who participated mainly to achieve some end goal; (b) activity-oriented, who participate for social or fellowship reasons; (c) learning-oriented, who perceive of learning as an end in itself. The results of this study reinforce the findings of Houle (1961). Study participants provided a variety of reasons for preparing and maintaining a teaching portfolio. Two thirds of the participants stated that they decided to prepare their teaching portfolio for award nominations, job applications, annual reviews, for promotion, or tenure process. This would suggest that these participants had, for the process of preparing a teaching portfolio, demonstrated characteristics of the goal-oriented learners as described by Houle (1961). Additionally, there were other participants who exhibited goal-oriented characteristics but in different ways. These participants explained that they prepared their teaching portfolios as a way to improve their teaching. The goal for these participants may not be summative in nature, but the goal of improved teaching skills and abilities is the reason for teaching portfolio preparation. In contrast, none of the participants stated that they prepared their teaching portfolios as a learning experience as an end in itself or for social reasons. It appears that faculty teaching portfolio preparation is done, primarily, for goal-oriented reasons.

These results have implications for faculty and postsecondary institutions who intend to use teaching portfolios for summative and/or formative evaluation when viewed through the lens of Houle's (1961) work. It is essential that faculty are aware of the various uses of teaching portfolios as an end goal of teaching portfolio preparation. This may motivate goal-oriented faculty to prepare a teaching portfolio. However, it is important to note that Houle (1961)

explained that while his participants generally fit within one of his three categories of continual learners, they also exhibited characteristics of the other two categories. As a result, it may be advisable for postsecondary institutions to provide teaching portfolio support and resources intended to attract learners who may be activity-oriented or learning-oriented.

Teaching Portfolio Preparation and Maintenance

The importance of providing support for faculty who are preparing a teaching portfolio has been noted by Seldin (1997). Seldin (1997) indicated that it is important to provide mentorship for those preparing a teaching portfolio in order to guide its construction and to ensure all relevant components are included. This study asked participants to reflect on the support and resources that were available to them as they prepared their teaching portfolio. The participants responded that they had accessed a wide variety of support and resources when preparing their teaching portfolio. They also identified which sources of support and resources they found helpful, as well as recommending possible resources that may assist other faculty who are in the initial stages of preparing a teaching portfolio. While it is difficult to conclude which sources of support are most valuable based on the results of this study, it is clear that a wide variety of resources are valued by faculty when preparing a teaching portfolio.

It is important postsecondary institutions provide varied support for faculty who are preparing a teaching portfolio. This support should come from written resources, examples, workshops, mentorship, teaching portfolio research, etc. Additionally, faculty should be aware of these resources so they can choose which ones they would like to use to assist them in preparing their self-directed teaching portfolio.

Teaching Portfolio Content

Scholars have noted that the content of teaching portfolios can vary significantly between faculty (Burns, 1999; Corry & Timmins, 2009; Simpson, Morzinski, & Lindemann, 1994; Urbach, 1992) and is dependent on whether the portfolio is prepared for formative or summative evaluation (Murray, 1995; Oermann, 1999). Additionally, teaching portfolio content is usually unique to the faculty member who prepares it (Reece et al., 2001; Seldin, 2000). The results of this study reinforce these positions as demonstrated by the responses of the participants.

Participants indicated that they used a wide variety of components for their teaching portfolios, that these components were most often selected by the participants, and, in some cases, the content of their teaching portfolio changed depending on usage. There were vast differences in the components that the participants included in their teaching portfolios; however, most participants included components that could be placed in the following categories: teaching philosophy, teaching responsibilities, professional development, evaluations, service, and research. This information adds to the body of research on teaching portfolio content by supporting the literature of other scholars (Berk 2005; Buckridge, 2008; CAUT, 2007; Corry & Timmins, 2009; Dinham & Scott, 2003; Kaplan, 1998) who have already created similar lists of possible teaching dossier components. It also adds to our knowledge of teaching dossier content by establishing that faculty have used components that are either different or titled differently than what was has been indicated by other scholars.

It is noteworthy that existing literature suggests that a faculty member's teaching philosophy be included in a teaching portfolio (Corry & Timmins, 2009; Murray, 1995; Reece et al., 2001; Seldin, 2010). While a majority of participants (18 of 21 participants) had included a teaching philosophy statement (or something similar but titled differently) in their teaching

portfolio, not all participants stated they had a teaching philosophy. This differs slightly from the existing literature and may indicate that the inclusion of a teaching philosophy statement is not perceived as a relevant component of a teaching portfolio.

Teaching portfolios are a SDL activity in which faculty have the agency to select the content of their teaching portfolio. The implications of these findings are that faculty may now be aware of more items from which to choose from to personalize their teaching portfolio. This may enhance faculty teaching portfolio use as a form of SDL.

Teaching Portfolio Uses

Scholars have indicated that teaching portfolios have been used by faculty for numerous summative reasons, including confirmation of tenure (Buckridge, 2008), promotion (Buckridge, 2008; Klenowski, Askew, & Carnell, 2006), award applications (Buckridge, 2008; Knapper, 1995), and teaching assessments (Klenowski, Askew, & Carnell, 2006). Additionally, the literature has noted that teaching portfolios can also be used for formative purposes (Dinham & Scott, 2003; Kaplan, 1998; Klenowski, Askew, & Carnell 2006; Knapper, 1995; Murray, 1995, Reece et al., 2001; Retallick, 2000; Weeks, 1996). Seldin, Miller and Seldin (2010) have stated that "by far, though, the two most often cited reasons for preparing teaching portfolios are to provide evidence for use in personnel decisions and to improve teaching performance" (p. 6). The results of this study confirm this, and indicate that faculty use teaching portfolios for various summative and formative purposes.

All of the participants indicated that they had used their teaching portfolio for summative purposes, such as annual reviews, promotion, tenure, job application, and/or award nomination. Participants also noted that they used their teaching portfolios for a variety of formative reasons. It is important that faculty are aware of the many uses of teaching portfolios.

The Perceived Effectiveness of Teaching Portfolio Use

Many scholars have indicated that teaching portfolios are effective for summative evaluation (Berk, 2005; Buckridge, 2008; Knapper, 1995; Weeks, 1996). Seldin, Miller and Seldin (2010) support the use of teaching portfolios for faculty summative evaluation:

The portfolio provides evaluators with hard-to-ignore information on what individual professors do as teachers, why they do it, how they do it, and the outcome of what they do.

And by so doing, it avoids looking at teaching as a derivative of student ratings. (p. 6) This perspective is also echoed by Murray (1995) who noted that teaching portfolios provide chairpersons and professors with the information needed to assess teaching. The majority of participants of this study (17 out of 21 participants) agreed that teaching portfolios can be an effective tool for summative evaluations. The reasons given for this position support much of the aforementioned literature on teaching portfolios for summative evaluations – that they provide a framework for faculty to provide evidence of their teaching.

While much of the literature on the summative use of faculty teaching portfolios suggests they are effective for such uses, it is not unanimous. Scholars have questioned the authenticity of teaching portfolios used for summative evaluation and have suggested that it is possible for faculty to provide information that may not be true to one's teaching (Bullard, 2000; Trevitt & Stocks, 2012). Additionally, there have been concerns in the literature that it may be difficult to assess faculty teaching portfolios as there is a lack of consistent criteria from which to assess them (Centra, 2000; Trevitt & Stocks, 2012) and, due to the diverse nature of teaching portfolios, it is very difficult to compare different teaching portfolios (Wright et al., 1999). Lastly, Wright et al. (1999) indicated that some faculty may find preparing a teaching portfolio a time-

consuming and burdensome task. These concerns were echoed by some of the participants of this study.

Study participants suggested that the value of teaching portfolios for summative evaluation may be limited by the accuracy of their content. The limited ability of an assessor to verify the content of a teaching portfolio may result in faculty providing false or omitting potentially damaging information. In addition to concerns regarding the truthfulness of teaching portfolio content, a few participants indicated that evaluating teaching portfolios objectively may be problematic. Time constraints may not allow for assessors to be adequately trained to evaluate teaching portfolios – particularly across disciplines - or to assess them thoroughly even if they are trained. Given that teaching portfolios are used extensively for summative evaluations, including tenure applications at numerous universities across Canada (Gravestock, 2011), it is important that these two points are addressed.

The implications regarding the effectiveness of teaching portfolios are numerous. While there is general agreement in the literature, including the results of this study, that teaching portfolios can be effective for summative evaluation of faculty, this may be contingent on a few key factors. Firstly, the validity of the content of faculty teaching portfolios used for summative evaluation needs to be considered. Postsecondary institutions would be wise to develop a process whereby teaching portfolio content can be verified. Secondly, the process whereby a teaching portfolio is evaluated should be standardized by the institution. This can be accomplished by developing a rating system that assessors can use and by making teaching portfolio content consistent for summative evaluations, campus-wide. Supporting this point, Seldin, Miller and Seldin (2010) noted

[P]ortfolios being used for personnel decisions such as tenure or promotion include certain items along with those chosen specifically by the professor...if certain items in the portfolio are standardized, comparisons of teaching performance (three finalists from different disciplines competing for university teacher of the year, for example) becomes possible. (pp. 6-7)

Although the use of teaching portfolios for faculty summative evaluations has challenges, there are avenues by which these evaluations can be made more effective.

In the literature there is mixed support for teaching portfolios as an effective tool for summative evaluations. The data in this study shows similarly mixed support, given certain contexts. While this study confirms many of the concerns in the literature, it presents a view that, for the majority of faculty, teaching portfolios are effective for summative evaluations. Additionally, it appears that there may be ways in which some of the aforementioned concerns can be alleviated.

Teaching Portfolios for Formative Evaluation

Teaching portfolios can be used for formative evaluation (Kaplan, 1998; Reece et al., 2001; Retallick, 2000). In fact, Seldin, Miller and Seldin (2010) suggested "there is no better reason to prepare a portfolio than to improve performance" (p. 8). Other scholars have also suggested that teaching portfolios can be effective for faculty formative evaluation (Buckridge, 2008; Dinahm & Scott, 2003; Seldin, 2000). While the results of this study revealed there are mixed views on the effectiveness of teaching portfolios for formative evaluation, overall, most participants in this study were supportive of the use of teaching portfolios for formative evaluations. While the reasons for this support are various, it appears that the reflective process associated with teaching

portfolio use, and the development of a teaching philosophy statement, have a perceived positive influence on the formative development of faculty teaching.

The reflective process that occurs with teaching portfolio use has been noted by many scholars as being particularly beneficial (Dinham & Scott, 2003; Kaplan, 1998; Knapper, 1995; Murray, 1995; Reece et al., 2001; Retallick, 2000; Seldin, 2000; Weeks, 1996). Seldin, Miller and Seldin (2010) explained how this occurs:

It is in the very process of reflecting on their work and creating a collection of documents and materials that professors are stimulated to reconsider policies and activities, rethink strategies and methodologies, revise priorities, and plan for the future. (p. 8)

Participants of this study frequently described how their teaching portfolios were a catalyst for reflection. The reflection that occurs through the preparation and maintenance of a teaching portfolio allowed faculty to revisit what they had done well and areas in which to improve, and then to use these insights to make improvements. This leads to what one participant appropriately termed the "cycle of improving teaching" (Participant 18A). This process of reflecting on teaching stimulates change and, ultimately, has a positive effect on teaching.

Teaching philosophy statements have been noted by scholars as being a key component of a teaching portfolio (Corry & Timmins, 2009; Murray, 1995; Reece et al., 2001; Seldin, Miller & Seldin, 2010). The development of a teaching philosophy statement was also noted by some study participants as a component of teaching portfolio preparation that has a positive formative effect on teaching. Preparing a teaching philosophy statement encouraged faculty to connect their beliefs about teaching with how they interact with their students and is sometimes used as a catalyst for change. Although not all participants indicated that their teaching portfolios included

a teaching philosophy statement, faculty who are in the initial stages of preparing a teaching portfolio would benefit from spending the time developing a teaching philosophy statement.

There are some scholars who question the effectiveness of teaching portfolios for formative purposes (Dinham & Scott, 2003; Wolf, 1996; Wright et al., 1999). Although the preparation and maintenance of a teaching portfolio may positively affect teaching, the results of this study indicate that this is inconclusive. There is a possibility that teaching improvements may not occur through the use of teaching portfolios, and it may be possible to see a greater effect on teaching through other formative methods. It is difficult to measure a cause and effect relationship between teaching portfolio use and the influence on teaching, so caution must be used if stating any definitive outcomes of teaching portfolio preparation and maintenance. This may hold to be an indefinite concern as teaching portfolios are a SDL activity that are by nature very different from one faculty member to the next. As such, assertions as to their effect are bound to be fraught with difficulties.

Teaching portfolios may not be effective for formative evaluation if faculty produce one for summative purposes only and are not using them to reflect on their teaching. As such, it is fair to question whether they can have a positive effect on teaching if they are merely a requirement for an award, yearly evaluations, tenure applications, or other summative evaluations. As a SDL activity, teaching portfolios may have a greater positive effect on teaching when internal motivation is a main reason for their preparation rather than to fulfil an obligation.

While uncertainty exists as to the value of teaching portfolios as formative evaluation tool, it is useful to note that 16 of the 21 participants of this study indicated that they perceived teaching portfolios as effective for formative evaluation. This effect may vary among faculty

members and is impossible to measure with a high level of accuracy; however, it cannot be discounted. There is agreement among most study participants that teaching portfolios can improve teaching and should be encouraged as a SDL activity.

Chapter Summary

Continuing from the results presented in chapter 4, this chapter discussed the main conclusions of this study: 1) a wide range of support and resources are found to be valued by faculty when preparing a teaching portfolio, 2) teaching portfolios are a SDL activity that vary in content, 3) teaching portfolios are initially prepared for numerous reasons and have a variety of uses, and 4) the effectiveness of teaching portfolios as a summative and/or formative evaluation tool vary among faculty. The implications of these conclusions were also discussed. In the next and final chapter, researcher reflections will be presented as well as possible avenues for future research in this area.

Chapter Six: Reflections and Recommendations

Overview

This study was conducted from my desire to explore how faculty prepare and maintain teaching portfolios for professional development and/or for summative evaluations. As a college instructor, I am always searching for ways to become a better educator and to have a more meaningful impact on my students. Soon after I began my post-graduate degree, I discovered teaching portfolios and began creating my own. I found that preparing a teaching portfolio, in my perception, affected my teaching in a positive way. However, I was still not satisfied with my teaching portfolio – there must be a way to use it more effectively, I thought. I conducted a literature review of teaching portfolio use by faculty and was surprised to find that there was limited research on the topic. Particularly, there was minimal research exploring the process by which faculty prepare and maintain a teaching portfolio or the perceived effect a teaching portfolio has on teaching. This perplexed me – why would there be so little literature on a tool that was used so frequently for faculty summative and formative evaluations? This lack of research exploring teaching portfolio use prompted this study. I sought to learn more about the use of teaching portfolios to satisfy my own desire to improve my teaching and to inform others to do the same.

This generic qualitative study was designed to address a gap in the literature describing teaching portfolio preparation and maintenance. I sought to gather data on the process of preparing and maintaining a teaching portfolio by interviewing faculty so that they could share their experiences and perceptions. SDL theory was used to help frame how teaching portfolios are used by postsecondary faculty with the objective of providing a more complete description of teaching portfolios as a summative and formative evaluation tool. I assumed that, as a form of

SDL, faculty teaching portfolio use would be varied and so too would be the perception of their effects. These experiences and perceptions, I hoped, would answer my research questions and would narrow the gap I found in my literature review.

The results of my study indicated that the process of creating a teaching portfolio by faculty exhibit the characteristics of a SDL process. Additionally, faculty prepare teaching portfolios for a variety of reasons; however, many do so for what Houle (1961) would describe as "goal-oriented" reasons. As a self-directed learning activity, faculty teaching portfolios are prepared in various ways, contain a multitude of components, and are used for numerous summative and formative evaluations. The perceived effect that teaching portfolios have on teaching is inconsistent as well: some faculty perceive them as very beneficial while others question the effect they have.

There were a number of conclusions and recommendations that were made based on the results of this study. Faculty teaching portfolios are a SDL activity that vary in content and are prepared using a wide range of support and resources. Teaching portfolios are also prepared for numerous reasons and have a variety of uses, and the effectiveness of teaching portfolios as a summative and/or formative evaluation tool vary among faculty. These conclusions were used as a basis to form suggestions that could be used to improve the effectiveness of teaching portfolios for summative and formative evaluation.

Recommendations for Future Study

The proliferation of teaching portfolios as a tool used for summative and formative evaluation necessitates that we have a full understanding of how they are used and to what degree they affect teaching. As the preparation and maintenance of a teaching portfolio is a SDL process resulting in a product of varying content and format, it is difficult to explore their use in

all contexts. However, it would prove beneficial to study their use in more detail through an examination of disciplinary differences. The results of such a line of research could provide additional insights into how teaching portfolios are used within and across the disciplines.

Another line of research that would prove beneficial would be the exploration of how assessors evaluate teaching portfolios for summative purposes. Teaching portfolio literature and the results of this study indicate that there is concern as to how teaching portfolios are evaluated. Research into the training of assessors and teaching portfolio evaluation processes could inform the summative use of teaching portfolios. It is likely that the results of this research could improve how teaching portfolios are evaluated and, in turn, reassure faculty that teaching portfolios are a valid tool for summative evaluation.

Lastly, a more in depth study of the support and resources used by faculty when preparing and maintaining a teaching portfolio would prove fruitful. Participants of this study indicated that they used a variety of resources while preparing their teaching portfolio and noted that many of them were quite helpful. However, these results only touch the surface of the importance of these resources – the bigger question of *why* they were valuable is still to be determined. A fuller understanding of how faculty can be assisted when preparing a teaching portfolio could certainly be used to improve their value and, as a result, the effectiveness of teaching portfolios for formative and summative evaluation.

References

- Alberta Advanced Education and Technology. (2007). Roles and mandates policy framework for Alberta's publicly funded advanced education system. Retrieved from Alberta Advanced Education website: http://advancededucation.alberta.ca/media/133783/rmpf.pdf
- Barrett, H. C. (2007). Researching electronic portfolios and learner engagement: The REFLECT initiative. *Journal of Adolescent & Adult Literacy*, 50(6), 436–449. Retrieved from https://www.gale.com/c/literature-resource-center
- Berk, R. A. (2005). Survey of 12 strategies to measure teaching effectiveness. *International Journal of Teaching and Learning in Higher Education*, 17(1), 48–62. Retrieved from http://www.isetl.org/ijtlhe/
- Brockett, R. G., & Donaghy, R. G. (2011). Self-directed learning: The Houle connection.

 International Journal of Self-Directed Learning, 8(2), 1–10.
- Brockett, R. G., Stockdale, S. L., Fogerson, D. L., Cox, B. F., Canipe, J. B., Chuprina, L. A., & Chadwell, N. E. (2000). *Two decades of literature on self-directed learning: A content analysis*. Retrieved from ERIC database. (ED449348)
- Brookfield, S. (1985). Self-directed learning: A critical review of research. *New Directions for Adult and Continuing Education*, 1985(25), 5–16.

 http://dx.doi.org/10.1002/ace.36719852503
- Brookfield, S. D. (2009). Self-directed learning. In R. Maclean & D. Wilson (Eds.), *International handbook of education for the changing world of work: Bridging academic and vocational learning* (pp. 2615–2627). New York, NY: Springer Netherlands.

- Buckridge, M. (2008). Teaching portfolios: Their role in teaching and learning policy.

 *International Journal for Academic Development, 13(2), 117–127. Retrieved from http://icedonline.net/international-journal-for-academic-development-ijad/
- Burns, C. W. (1999). Teaching portfolios and the evaluation of teaching in higher education:

 Confident claims, questionable research support. *Studies in Educational Evaluation*,

 25(2), 131–142. Retrieved from https://doi.org/10.1016/S0191-491X(99)00017-6
- Caelli, K., Ray, L., & Mill, J. (2003). 'Clear as mud': Toward greater clarity in generic qualitative research. *International Journal of qualitative methods, 2*(2), 1–13. Retrieved from http://journals.sagepub.com/home/ijq
- Caffarella, R. S. (1993). Self-directed learning. New Directions for Adult Learning and Continuing Education, 1993(57), 25–35. http://dx.doi.org/10.1002/ace.36719935705
- CAUT. (2007). *Teaching dossier*. Retrieved from Canadian Association of University Teachers website: https://www.caut.ca/docs/default-source/professional-advice/teaching-dossier.pdf?sfvrsn=2
- Centra, J. A. (2000). Evaluating the teaching portfolio: A role for colleagues. *New Directions for Teaching and Learning*, 2000(83), 87–93. Retrieved from http://onlinelibrary.wiley.com/journal/10.1002/(ISSN)1536-0768
- Confessore, G. J., & Confessore, S. J. (1992). Selecting the key literature in self-directed learning. In G. J. Confessore & S. J. Confessore (Eds.), *Guideposts to self-directed learning: Expert commentary on essential concepts* (pp. 12–22). King of Prussia, PA: Organization Design and Development.

- Confessore, S. J., & Kops, W. J. (1998). Self-directed learning and the learning organization:

 Examining the connection between the individual and the learning environment. *Human Resource Development Quarterly*, *9*(4), 365–375.

 http://dx.doi.org/10.1002/hrdq.3920090407
- Corry, M., & Timmins, F. (2009). The use of teaching portfolios to promote excellence and scholarship in nurse education. *Nurse Education in Practice*, *9*(6), 388–392. http://dx.doi.org/10.1016/j.nepr.2008.11.005
- Dinham, S., & Scott, C. (2003). Benefits to teachers of the professional learning portfolio: A case study. *Teacher Development*, 7(2), 229–244. Retrieved from http://www.tandfonline.com/doi/pdf/10.1080/13664530300200200
- Garrison, D. R. (1997). Self-directed learning: Toward a comprehensive model. *Adult Education Quarterly*, 48(1), 18–33. http://dx.doi.org/10.1177/074171369704800103
- Gravestock, P. S. (2011). Does teaching matter? The role of teaching evaluation in tenure

 policies at selected canadian universities (Doctoral dissertation, University of Toronto,

 Toronto, Canada). Retrieved from

 https://tspace.library.utoronto.ca/bitstream/1807/31764/6/Gravestock_Pamela_S_201111

 _PhD_thesis.pdf
- Guglielmino, L. M., Long, H. B., & Hiemstra, R. (2004). Self-direction in learning in the United States. *International Journal of Self-Directed Learning*, *I*(1), 1–17. Retrieved from http://sdlglobal.com/journals.php
- Hiemstra, R. (1994). Self-directed learning. In T. Husen & T. N. Postlethwaite (Eds.), *The* international encyclopedia of education (2nd ed.). Oxford, England: Pergamon Press.

- Houle, C. O. (1961). *The inquiring mind*. Madison, WI: The University of Wisconsin Press.
- Kaplan, M. (1998). The teaching portfolio. *CRLT Occasional Paper*, *11*, 1–7. Retrieved from https://web.uri.edu/teach/files/Teaching-Portfolio.pdf
- Klenowski, V., Askew, S., & Carnell, E. (2006). Portfolios for learning, assessment and professional development in higher education. *Assessment & Evaluation in Higher Education*, 31(3), 267–286. http://dx.doi.org/10.1080/02602930500352816
- Knapper, C. K. (1995). The origins of teaching portfolios. *Journal on Excellence in College Teaching*, 6(1), 267–286.
- Knowles, M. S. (1975). *Self-directed learning: A guide for learners and teachers*. New York, NY: Association Press.
- Leavy, P., & Hesse-Bieber, S. N. (2006). *The practice of qualitative research*. Thousand Oaks, CA: SAGE Publications.
- Manning, K. (2013). Organizational theory in higher education. New York, NY: Routledge.
- McLean, M., & Bullard, J. (2000). Becoming a university teacher: Evidence from teaching portfolios (how academics learn to teach). *Teacher Development*, 4(1), 79–101. http://dx.doi.org/10.1080/13664530000200104
- Merriam, S. B. (2001). Andragogy and self-directed learning: Pillars of adult learning theory.

 *New Directions for Adult and Continuing Education, 2001(89), 3–14.

 http://dx.doi.org/10.1002/ace.3

- Mertens, D. M. (2010). Research and evaluation in education and psychology: Integrating diversity with quantitative, qualitative, and mixed methods. Thousand Oaks, CA: SAGE publications.
- Mezirow, J. (1985). A critical theory of self-directed learning. *New Directions for Adult and Continuing Education*, 1985(25), 17–30. http://dx.doi.org/10.1002/ace.36719852504
- Murray, J. P. (1995). The teaching portfolio: A tool for department chairpersons to create a climate of teaching excellence. *Innovative Higher Education*, 19(3), 163–175. Retrieved from https://doi.org/10.1007/BF01191217
- Oermann, M. H. (1999). Developing a teaching portfolio. *Journal of Professional Nursing*, 15(4), 224–228. http://dx.doi.org/10.1016/S8755-7223(99)80008-8
- Patton, M. H. (1990). *Qualitative evaluation and research methods* (2nd ed.). Thousand Oaks, CA: SAGE Publications.
- Reece, S. M., Pearce, C. W., Melillo, K. D., & Beaudry, M. (2001). The faculty portfolio:

 Documenting the scholarship of teaching. *Journal of Professional Nursing*, *17*(4), 180–186. http://dx.doi.org/10.1053/jpnu.2001.24861
- Retallick, J. (1997). *The portfolio guide* (2nd ed.). Sydney, Australia: Association of Consulting Engineers Australia and NSW Department of Education and Training.
- Schönwetter, D. J., Sokal, L., Friesen, M., & Taylor, K. L. (2002). Teaching philosophies reconsidered: A conceptual model for the development and evaluation of teaching philosophy statements. *International Journal for Academic Development*, 7(1), 83–97. http://dx.doi.org/10.1080/13601440210156501

- Seldin, P. (1997). *The teaching portfolio: A practical guide to improved performance and promotion/tenure decisions*. Bolton, MA: Anker Publishing Company, Inc.
- Seldin, P. (2000). Teaching portfolios: A positive appraisal. *Academe*, 86(1), 36. http://dx.doi.org/10.2307/40252334
- Seldin, P., Miller, J. E., & Seldin, C. A. (2010). The teaching portfolio: A practical guide to improved performance and promotion/tenure decisions (4th ed.). Hoboken, NJ: Wiley.
- Simpson, D., Morzinski, J., Beecher, A., & Lindemann, J. (1994). Meeting the challenge to document teaching accomplishments: The educator's portfolio. *Teaching and Learning in Medicine: An International Journal*, *6*(3), 203–206. Retrieved from https://doi.org/10.1080/10401339409539677
- Statistics Canada. (2010). Definition of postsecondary education. Retrieved from http://www.statcan.gc.ca/pub/81-004-x/2010001/def/posteducation-educpost-eng.htm
- Tough, A. (1978). Major learning efforts: Recent research and future directions. *Adult Education*, 28(4), 250–263. Retrieved from https://doi.org/10.1177/074171367802800403
- Trevitt, C., & Stocks, C. (2012). Signifying authenticity in academic practice: A framework for better understanding and harnessing portfolio assessment. *Assessment & Evaluation in Higher Education*, 37(2), 245–257. http://dx.doi.org/10.1080/02602938.2010.527916
- University of Alberta. (2006). *University of Alberta faculty agreement*. Retrieved from Association of Academic Staff University of Alberta website: http://www.aasua.ca/wp-content/uploads/2013/06/FacultyAgreement amendedNov2010.pdf

- Urbach, F. (1992). Developing a teaching portfolio. *College Teaching, 40*(2), 71–74. Retrieved from http://www.jstor.org/stable/27558526
- Weeks, P. A. (1996). The teaching portfolio: A professional development tool. *The International Journal for Academic Development*, *I*(1), 70–74. Retrieved from https://doi.org/10.1080/1360144960010108
- Wolf, K. (1996). Developing an effective teaching portfolio. *Educational Leadership*, *53*(6), 34–37. Retrieved from http://www.ascd.org/publications/educational-leadership.aspx
- Wright, W. A., Knight, P. T., & Pomerleau, N. (1999). Portfolio People: Teaching and Learning Dossiers and Innovation in Higher Education. *Innovative Higher Education*, 24(2), 89–103. Retrieved from https://doi.org/10.1023/B:IHIE.0000008148.71650.e6

Appendix A: Department Chair Email

(Date)

Dear (name of department chair)

This letter is to inform you of a research study I will be undertaking as part of my master's program and to request your assistance in the recruitment of potential research participants. The following is a brief description of the research study and the participant recruitment involvement I am requesting of yourself.

- The **purpose of this study** is to add to the body of knowledge regarding the preparation and maintenance of teaching portfolios by university faculty members.
- Research participants will take part in confidential one-on-one interviews which should take approximately 30-60 minutes. Interview questions will explore the experiences that faculty members have had preparing a teaching portfolio, maintaining a teaching portfolio, and whether or not they think that their teaching portfolio has enhanced their teaching abilities.
- Due to your access to confidential contact information I am requesting your assistance in the participant recruitment process. I hope that you would be willing to forward this email and the attached "Letter of Invitation" to faculty members of your department for their consideration. Subsequently, any interested faculty members can contact myself directly for more information or to participate in the study.

Thank you for your time and consideration.

Sincerely,

Byron Matthews
University of Alberta
Faculty of Education
Department of Educational Policies
Masters of Education Student
bwmatthe@ualberta.ca
780-242-6866

Thesis Supervisor

Heather Kanuka

Heather.kanuka@ualberta.ca

University of Alberta

Faculty of Education

Department of Educational Policies

Professor

Appendix B: Letter of Invitation

(Date)

Dear Potential Study Participant,

I'm a master's student in the Educational Policy Studies department in the Faculty of Education. I would like to carry out a study that involves interviewing approximately 30 faculty members exploring faculty experiences with the preparation and maintenance of a teaching portfolio.

I realize that there are many demands on your time and that you may not be able to commit to this at this time. However, the study would greatly benefit from your participation and sharing of your experiences. Information collected in the interviews will be compared and contrasted in order to enhance our understanding of the effectiveness of preparing and maintaining a teaching portfolio.

Purpose of Study

The purpose of this research is to add to the body of knowledge regarding the preparation and maintenance of teaching portfolios by university faculty members. Currently there is limited research exploring the experiences of faculty members when preparing and maintaining a teaching portfolio. This study will provide an in depth analysis of these experiences of a number of University of Alberta faculty members from a variety of backgrounds. The results of this study will be primarily used in support of my masters of education thesis but will also be used for conference presentations and scholarly publications.

Data Analysis: Interview transcriptions will be analyzed using the constant comparison method in Grounded Theory, followed by looking for themes and topics that emerge (with debriefing with my thesis supervisor). Following an initial analysis, the themes and topics will be returned to you for a member check for verification of content. You will then have one month to review the content, to approve it and return it to the researcher, or to request revisions. Once the approved transcriptions have been returned their content will be analyzed in order to find similar or contrasting responses.

Here are some other details of the study:

- Interviews will consist of 25-40 semi-structured questions.
- Interviews will take approximately 30-60 minutes.
- Interviews will be scheduled at your convenience.

Participation in this study does not involve any risks greater than those encountered in everyday life. You are free to refuse to answer particular questions, to quit the interview, or to withdraw from the study altogether prior to submitting the interview data. Once the data has been submitted it cannot be remove as all identifying personal information will be removed so that it is anonymous. Data will be stored on a password-protected hard drive in a secure office. The data will be retained for five years, per University policy. There is no remuneration or compensation for participating in this study.

If you are willing to participate in this study please contact myself directly (bwmatthe@ualberta.ca) to arrange a date and time for this interview with me. At the onset of the interview, I will ask you to read the description of the study and sign a consent form.

Thank you for considering my invitation to participate. If you have any questions or concerns regarding your rights as a participant, or how this study is being conducted, you may contact the University of Alberta's Research Ethics Office at 780-492-2615. This office has no affiliation with the study investigators.

Sincerely,

Byron Matthews bwmatthe@ualberta.ca University of Alberta Faculty of Education Department of Educational Policies Masters of Education Student

Appendix C: Participant Consent

Study Title: A study of the experiences of teaching portfolio preparation and maintenance by University of Alberta faculty

This consent has two parts. The first part is a description of the study and your role as a participant. The second part is the consent itself. Reading the description and signing the consent to participate indicates that you are giving your informed consent to participate in the study. If you would like more information about the study or the consent process please feel free to email me bwmatthe@ualberta.ca

1. Description of the study

Purpose and Objectives of the Study

The purpose of this research is to add to the body of knowledge regarding the preparation and maintenance of teaching portfolios by university faculty members. Currently there is limited research exploring the experiences of faculty members when preparing and maintaining a teaching portfolio. This study will provide an in depth analysis of these experiences of a number of University of Alberta faculty members from a variety of backgrounds.

Research Design

Selection of study participants. Approximately 30 faculty members of the University of Alberta who have prepared a teaching portfolio will be interviewed for the study.

Recruitment of study participants. Participation in the study is voluntary. Participants will be invited to participate via an email invitation distributed by University of Alberta department chairs. Department chairs will be informed of the study via email and will be asked to support the study by distributing the participant request email to faculty members in their department. Interested participants will be asked to contact the researcher directly for more information or to participate in the study.

Data collection and analysis.

- The study will require you to participate in a one-on-one interview with the researcher prior to July 1st, 2015.
- The interview will consist of 25-40 open-ended questions and will take approximately 30-60 minutes to complete.
- The interview will take place in a quiet, private room at the University of Alberta.
- You will be asked to email the researcher with dates and times that are convenient for you to do the interview. Interview date, time, and location will be then confirmed by the researcher. If you need to reschedule the interview for any reason you will be able to do so as long as it is prior to July 1st, 2015.

- Only you and the researcher will be present during the interview.
- The interview will be recorded using a digital voice recording device for later transcription by the researcher.
- The transcriptions will be returned to you within one month following your interview for verification of content. You will have one month to review the transcription, to approve it and return it to the researcher, or to request revisions.
- You will be provided with the results of the study after all the interviews have been completed and analyzed. It is anticipated that this will occur prior to the end of 2015.

Reporting of the results. The results of this study will be primarily used in support of my masters of education thesis but will also be used for scholarly presentations and academic publications. The results will also be provided via email to all individuals who have participated at the end of March, 2016.

Risks. Participation in this study does not involve any risks greater than those encountered in everyday life.

Freedom to withdraw. Participants are free to refuse to answer any (or all) questions, to quit the interview, or to withdraw from the study altogether. Once data has been analyzed the participant cannot have his or her information removed.

Other Considerations

Data will be stored on a password-protected hard drive in a secure office. The data will be retained for five years, per University policy. There is no remuneration or compensation for participating in this study.

The plan for this study has been reviewed for its adherence to ethical guidelines by a Research Ethics Board at the University of Alberta.

2. Consent

By participating in the study, in no way does this waive your legal rights nor release the investigators, sponsors, or involved institutions from their legal and professional responsibilities. You are free to withdraw from the study at any time prior to the data analysis. Your continued participation should be informed as your initial consent, so you should feel free to ask for clarification or new information throughout your participation.

Your signature below indicates that you would like to participate in the interview, and will constitute your consent.

Signature of Research Participant	Date	
Witness	Date	

Appendix D: Interview Questions

Part 1: Demographics

- 1) What department do you work in?
- 2) How long have you been teaching in a university? Where else have you taught? How long?
- 3) Do you have any teacher training or preparation? Was this formal or informal? Credit or non-credit?
- 4) Are you currently in a teaching-stream or research-stream?

Part 2: Teaching Portfolio Preparation and Maintenance

- 1) How did you first learn about teaching portfolios?
- 2) When did you first prepare a teaching portfolio?
- 3) Why did you decide to prepare a teaching portfolio?
- 4) What forms of guidance or resources (ie. workshop, mentorship) were available to you when you first prepared your teaching portfolio? Was this helpful? What would you change to make the process more effective?
- 5) How much time did you spend on preparing your teaching portfolio? Was this enough time? Why/Why not?
- 6) What components does your teaching portfolio consist of? Why did you choose them?
- 7) Which components do you find most helpful?
- 8) Which components do you find least helpful?
- 9) What purposes have you used your teaching portfolio for? (ie. summative evaluation, formative evaluations, awards, grants, job applications)
- 10) Is your teaching portfolio online or a hard copy? Why did you choose that format? Sub-Question "4A": (If the participants teaching portfolio is a hard copy) Would you consider preparing an e-portfolio?
 - Sub-Question "4B": (If the participant answers NO to 4A) Why not? Sub-Question "4C": (If the participant answers YES to 4A) What forms of guidance or resources would you want to assist you in preparing an e-portfolio?
- 11) How often do you update your teaching portfolio?
- 12) Have you added or removed components while updating your teaching portfolio? Why?
- 13) Are faculty members in your department encouraged to prepare or maintain a teaching portfolio? In what ways?

Part 3: Teaching Portfolio Effectiveness

- 1) Has the process of preparing and maintaining a teaching portfolio affected your teaching? In what ways/Why not?
- 2) Do you think teaching portfolios improve teaching? In what ways/Why not?
- 3) Do you think teaching portfolios are effective for summative evaluation? Why/Why not?
- 4) Do you think teaching portfolios are effective for formative evaluation? Why/Why not?
- 5) Do you think teaching portfolios can be effectively used for both summative and formative evaluation? Why/Why not?

- 6) Do you think that preparing and maintaining a teaching portfolio is an effective use of time? Why/Why not?
- 7) Would you recommend preparing and maintaining a teaching portfolio to a colleague who does not have one? Why/Why not?